

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Schüwer, Ulrich; Kosfeld, Michael

Conference Paper Add-on Pricing, Consumer Myopia and Regulatory Intervention

Beiträge zur Jahrestagung des Vereins für Socialpolitik 2010: Ökonomie der Familie - Session: Regulation and Pricing Policies, No. F8-V2

Provided in Cooperation with:

Verein für Socialpolitik / German Economic Association

Suggested Citation: Schüwer, Ulrich; Kosfeld, Michael (2010) : Add-on Pricing, Consumer Myopia and Regulatory Intervention, Beiträge zur Jahrestagung des Vereins für Socialpolitik 2010: Ökonomie der Familie - Session: Regulation and Pricing Policies, No. F8-V2, Verein für Socialpolitik, Frankfurt a. M.

This Version is available at: https://hdl.handle.net/10419/37278

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Add-on Pricing, Consumer Myopia and Regulatory Intervention

Michael Kosfeld*

Ulrich Schüwer[†]

July 2010

Abstract

Previous research shows that firms may offer excessively high (and shrouded) add-on prices in competitive markets when some consumers are myopic. We analyze the effects of regulatory intervention via educating myopic consumers on equilibrium strategies of firms, consumer protection and welfare. Our model allows firms to price discriminate between consumers. We show that intervention may actually decrease social welfare and weaken consumer protection if regulatory efforts are insufficient to alter the equilibrium strategy of firms. In consequence, regulation may do more harm than good and hence should only be considered if the regulator can be sure to have a good understanding of pricing strategies in an industry.

Keywords: bounded rationality, information suppression, regulation, social welfare, consumer protection JEL Classification: D40, D80, L50

^{*}Goethe-University Frankfurt, Department of Economics and Business Administration, Grüneburgplatz 1, 60323 Frankfurt am Main, Germany, E-mail: kosfeld@econ.uni-frankfurt.de.

[†]Goethe-University Frankfurt, Department of Finance, Grüneburgplatz 1, 60323 Frankfurt am Main, Germany, E-mail: schuewer@finance.uni-frankfurt.de.

1 Introduction

When consumers are boundedly rational and myopic in their decision making, firms will try to exploit it. Well-known examples are that consumers book a hotel room without considering the costs of the minibar, buy a computer without considering the costs for upgrades, or open a bank account without considering fees for additional banking services. As shown by Gabaix and Laibson (2006), the equilibrium pricing strategy of firms then may be to compete purely on the price of a base good (e.g., a hotel room, computer without upgrades, or bank account) and to suppress information about add-ons (e.g., the minibar, computer upgrades, or banking fees). The firm picks a price for the base good below marginal costs, and a price for the add-on above marginal costs. Consequences for consumers are twofold: First, sophisticated consumers substitute away from add-ons, which creates a social welfare loss if substitution is costly. Second, myopic consumers who buy the high-priced add-on subsidize sophisticated consumers, which raises consumer protection concerns. The question arises if and how the regulator should intervene to maximize social welfare and consumer protection.

In many situations, vigorous competition is the single best instrument for efficient market outcomes and for protecting consumer interests, and regulation is not needed (e.g., Armstrong, 2008). Regulation and consumer protection policies may even have possibly undesirable effects if it leads to a weakening of competitive pressures (e.g., Armstrong, Vickers, and Zhou, 2009). In some specific situations, however, regulatory intervention can help markets function better. For example, there can be a role for regulation to create transparency (Spiegler, 2006a,b). If regulatory intervention benefits vulnerable consumers without harming more sophisticated consumers, it is relatively uncontroversial. If interventionist consumer policies benefit one group of consumers at the expense of another, it is not so clear if regulatory intervention is warranted (Armstrong, 2008). Moreover, regulatory intervention is only relevant in fields where the regulator cares about the decisions of uninformed consumers. Uninformed consumers are not in general those citizens who need public subsidies. For example, the regulator probably does not care about add-on prices of luxury cars, but cares about high add-on prices in the context of tax-subsidized retirement funds (otherwise, why the tax subsidies in the first place?). Efficient regulatory intervention is challenging.¹ An obvious regulatory intervention is to require firms to inform consumers about all relevant product characteristics and prices. Alternatively, the regulator could impose price caps on overpriced add-ons. Both strategies could work well when the firm offers one base good and one or few add-ons, e.g., a bank account and overdraft fees. In practice, however, the products and/or services offered are often much more complex. For example, a computer is offered with many potential upgrades, and a bank account comes with many complementary financial services. Thus, when certain products are regulated, firms may find ways to circumvent such regulation.

This paper analyzes regulatory intervention when firms exploit consumers who are myopic in their decision making. A regulator may educate myopic consumers about add-on prices before firms decide on their pricing strategy. Such regulatory intervention could be through general public information or requiring "warning labels" placed on marketing materials. The rationale for such an intervention is that *shrouding* information about add-ons only prevails for a sufficiently high portion of uninformed myopic consumers, and that *unshrouding*, which is socially desirable, prevails otherwise. Such efforts to correct individuals' factual mistakes are largely uncontroversial in politics and the academic literature (see, e.g., Jolls and Sunstein, 2006, 230,232). What is not so clear is to which degree educating consumers can change the behavior of myopic consumers. Recent research finds that a surprisingly small fraction of consumers makes better decisions after being informed about important decision criteria.² Thus, it is unlikely that a regulator can make all consumers act like sophisticated consumers through education, and a more or less large fraction of uninformed consumers will remain. Furthermore, we consider that firms may own a technology that allows them to price discriminate between myopic and sophisticated consumers (more or less successfully). This may lead to new price equilibria, and mitigate the social welfare problem even without regulatory intervention. However, it may also have adverse effects, and it will not solve potential consumer protection concerns. Effects of regulatory intervention therefore also depend on the potential of firms to price discriminate. Finally, we analyze alternative ways of regulatory intervention

 $^{^{1}}$ Gabaix and Laibson (2006, 530-31) discuss verbally alternative remedies that the regulator can consider, but do not derive formal results.

²For a literature review about the limited impact of financial education on consumer behavior, see De Meza, Irlenbusch, and Reyniers (2008). Choi, Laibson, and Madrian (2010) and Beshears, Choi, Laibson, and Madrian (2009) provide experimental evidence regarding investments in mutual funds.

through subsidizing firms that accept to follow a certain (loss-making) pricing strategy that is in the public interest.

Generally, price discrimination exists when the difference in prices among consumers is not proportional to the difference in marginal costs (see, e.g., Stole, 2007). We consider price discrimination that is based on observed consumer heterogeneity (e.g., past purchasing decisions, age, profession, education), called *third-degree* or *direct* price discrimination: The firm offers add-on products at different prices to consumers who the firm considers to be myopic or sophisticated.³ Potential profits of firms from charging different prices to different consumers are competed away in our model. However, the potential to price discriminate impacts the firms' pricing strategies, and thereby consumers' buying decisions, social welfare and consumer protection.

To illustrate, consider the following example. A bank offers a cash account (the base good) and one or two investment funds (the add-ons). The first fund is a high-priced managed investment fund and the second fund is a low-priced exchange traded fund (ETF). Both funds track the same index and can be considered as perfect substitutes. Informed consumers consider both the price for the cash account and the price for the investment fund for their buying decision while uninformed consumers only consider the price of the cash account. The pricing strategy of the bank then involves two decisions. First, whether its range of products consists of the managed investment fund, the ETF, or both funds. Second, whether to compete purely on the price of the cash account, e.g., by offering high interest rates, or also to advertise the fees of the investment funds. If the bank owns a technology to price differentiate between myopic and sophisticated investors, the bank can shroud information about add-ons towards myopic consumers and offer them managed investment funds once they open a cash account. Furthermore, the bank can unshroud information about ETFs towards sophisticated consumers.⁴

³Alternatively, *first-degree* price discrimination exists when the firm charges a different price per unit of product and per consumer. *Second-degree* price discrimination exists when the firm charges different prices based on unobservable consumer heterogeneity, because consumers self-select into the appropriate product. Special forms of price discrimination are, e.g., bundling, introductory offers, poaching and versioning. For a literature review on price discrimination, see Armstrong (2006), Esteves (2009), Fudenberg and Villas-Boas (2006), or Stole (2007).

 $^{^{4}}$ ETFs are traded on the German stock exchange since 2000. However, until recently most banks in Germany offered only managed funds and no ETFs. This seems to change. Deutsche Bank, for example, offers ETFs under the name *db x-trackers* since 2007. The German savings banks offer ETFs under the name DEKA since

Another example in the field of retail finance are credit cards. As discussed by Ausubel (1991), consumers often make credit card choices without considering the high probability that they will pay high interest rates on their outstanding balances.

As a reference point for our analysis, consider the following situations: first, all consumers are informed, and second, all consumers are uninformed. In the first situation, the profit maximizing pricing strategy of firms is to compete on both the price of the base good and the price of the add-on. The bundle of both products will be offered at cost-reflective prices to all consumers. In the second situation, the profit maximizing pricing strategy of firms is to compete purely on the price of the base good and to charge the reservation price of consumers for the add-on. Again, the bundle of both products will be offered at cost-reflective prices to all consumers because the firms' profits from high add-on prices get competed away on the base good. Such a price equilibrium represent the so-called *loss-leader pricing* (Lal and Matutes, 1994). Thus, the consumer harm is limited to the inefficiency caused by the departure of prices from costs on a product-by-product basis (the "deadweight loss"), e.g., consumers replace the base good too early. It is generally argued that this welfare loss is relatively small (see, e.g., Armstrong, 2008, footnote 64; Shapiro, 1994, 497-498). In sum, as well a situation where all consumers are informed as a situation where all consumers are uninformed does in general not require regulatory intervention. The social welfare and consumer protection issues discussed in this paper arise because of *heterogenous* consumers.

Our main results are as follows. First, if educating consumers changes a high-priced add-on equilibrium or a price discrimination equilibrium of firms, social welfare increases. Second, if educating consumers does not change a high-priced add-on equilibrium or a price discrimination equilibrium, social welfare decreases. Finally, if a low-priced add-on equilibrium exists without educating consumers, intervention is obviously irrelevant because there is no social welfare loss in the first place. Effects depend on the fraction of myopic consumers, the efficiency of educating myopic consumers, the effort costs of substituting away from add-ons, the upper level for the add-on price, and the efficiency of price discrimination. We develop alternative regulatory strategies based on subsidizing firms, and we show that these strategies

^{2008.} An exception is, e.g., Citibank Deutschland (now called Targobank and part of Credit Mutuel Group) who does not offer own ETFs. Until 2008 Citibank Deutschland even did not allow customers to buy ETFs from external providers into their securities account, officially because of "technical difficulties".

can create a social welfare gain. In summary, the regulator needs to carefully analyze the situation before intervening via educating myopic consumers and may find no intervention at all or other regulatory strategies more beneficial.

The paper proceeds as follows. Section 2 contains the details of the model (2.1), the price equilibria when the regulator decides not to educate myopic consumers (2.2), and the price equilibria when the regulator decides to educate myopic consumers (2.3). Section 3 discusses implications for social welfare and consumer protection. Section 4 discusses alternative ways of regulatory intervention through subsidizing firms. Finally, Section 5 contains concluding remarks. Proofs and detailed calculations appear in the appendix.

2 Model

The following pricing game is closely related to the model of Gabaix and Laibson (2006). The innovation of our model is twofold. First, the model allows firms to choose price discrimination besides uniform pricing. Second, a regulator may intervene and educate myopic consumers about the relevance of add-on prices.

2.1 Details

Firms produce base goods and add-ons. Add-ons are always avoidable for informed consumers, in contrast to unavoidable *surcharges*. For simplicity, marginal costs for the base good and for add-ons are zero. The mass of consumers is normalized to 1. Many firms exist and, depending on the parameters of the demand function of consumers, a zero profit equilibrium may exist.

Consumer types. Two types of consumers are considered: myopic consumers, a fraction α of all consumers, where $\alpha \in (0, 1]$, and sophisticated consumers, a fraction $1 - \alpha$, of all consumers. Myopic consumers per se only take the price for the base good into consideration when deciding where to purchase a good. Sophisticated consumers always take both the price for the base good and the price for the add-on into consideration when deciding where to

purchase a good.

Information strategy of firms. Shrouding means that the firm suppresses information about the price of the add-on. Unshrouding means that the firm provides information about the price of the add-on. By providing this price, the firm at the same time educates a fraction λ_F of myopic consumers about the relevance of the add-on price, where $\lambda_F \in [0,1]$. Thus, this fraction of myopic consumers becomes informed and is therefore called *informed myopic*. Unshrouding is free.

Regulatory intervention through education. The regulator can educate a fraction $\lambda_R \in [0, 1]$ of myopic consumers about the relevance of add-on prices, which increases the fraction of informed consumers before firms decide on a pricing strategy from $1 - \alpha$ to $1 - (1 - \lambda_R)\alpha$. Educating consumers is free. Furthermore, a fraction $\lambda'_F \in [0, \lambda_F]$ of the remaining uninformed myopic consumers becomes informed when firms unshroud. At the one extreme when $\lambda'_F = 0$, it is assumed that following education through a regulator, unshrouding through firms has no complementary effect because all myopic consumers who potentially could become informed already got informed through the regulator. The remaining fraction of the population is just not receptive for any kind of information, or not capable to use any but the most simple decision rationale. At the other extreme when $\lambda'_F = \lambda_F$, it is assumed that unshrouding through firms always informs an additional fraction λ_F of the remaining uninformed consumers, regardless of any potential prior education through the regulator.⁵

Pricing strategy of firms. The firm chooses a price for the base good, p, and a price for the add-on, \hat{p} . In case of price discrimination, the prices of the low-priced add-on and the high-priced add-on are denoted as \hat{p}_L and \hat{p}_H . The prices for the add-ons have an upper bound, $\bar{p} \in (0, 1]$.⁶ No consumer will buy an add-on above this price. All prices refer to firm i (for simplicity, the subscript i is omitted). We focus on symmetric price equilibria.

⁵We consider alternative regulatory strategies, namely intervention through subsidized public firms, in an extension of the model in Section 4.

⁶The assumption that $\bar{p} \in (0, 1]$ simplifies Corollary 3, otherwise it does not matter.

Price discrimination. Firms may own a technology that allows them to use interpersonal information about consumers (previous purchases, education, etc.) in order to classify them as uninformed or informed consumer. Then, the firm may offer the add-on to uninformed and to informed consumers at different prices. The reservation price for the add-on of informed consumers is relatively low because they are in a position to substitute away. Thus, the firm offers them a low-priced add-on. The reservation price for the add-on of uninformed consumers is relatively high. Thus, the firm offers them a high-priced add-on. For simplicity, we assume that all add-ons have the same quality and consumer utility.⁷ If available, the technology for price discrimination is free. Realistically, price discrimination works more or less well, depending on the industry, product, and customer characteristics. We consider two potential limits:

- The error when firms classify an informed consumer as uninformed is called *attrition*. This error implies that the firm will unintentionally *shroud* (not advertise) the lowpriced add-on towards a fraction $1 - \beta$ of informed consumers. It follows that this fraction will substitute away.
- The error when firms classify an uninformed consumer as informed consumer is called *cannibalization*. This error implies that the firm will unintentionally *unshroud* (advertise) the low-priced add-on towards a fraction 1γ of uninformed consumers. It follows that these customers buy the low-priced add-on instead of the high-priced add-on.

See Figure 1 for an overview on effects of price discrimination.

Demand function. The demand function of firm *i* is modeled as the probability density function that a consumer purchases at firm *i*, $D(x_i) \in [0, 1]$, where x_i represents the *anticipated net surplus* from purchasing a product at firm *i* less the *anticipated net surplus* from purchasing a product at the best alternative firm. The demand function $D(x_i)$ is strictly increasing and reflects the degree of competition in the industry that is given by $\mu = \frac{D(0)}{D'(0)}$.

⁷A weaker assumption would be that prices of both add-ons are not proportional to the difference in marginal costs, the so-called *versioning* (see, e.g., Shapiro and Varian, 1998).

Perfect competition corresponds to $\mu = 0.^8$ The arguments of the decision rationale x_i depend on the consumer type that is either an *informed consumer* (sophisticated and informed myopic) or an *uninformed consumer*.

- For uninformed consumers, x_i is given by $\bar{x}_i = -p + p^*$, where p represents the price for the base good at firm i, and p^* represents the price for the base good at the best alternative firm. Uninformed consumers ignore prices of add-ons when they make their buying decision.
- For informed consumers, x_i is given by x̂_i = -p min{Ep̂, e} + p* + min{Ep̂*, e}, where p and p* are defined as above, Ep̂ represents the expected price for the add-on at firm i, Ep̂* represents the expected price for the add on at the best alternative firm, and e represents the costs for substituting away from the add-on. If firms unshroud, the expected prices Ep̂ and Ep̂* become the known prices p̂ and p̂*, respectively.

Timing. The model has four periods. In period 0 the regulator may intervene. In period 1 firms determine their pricing strategy. In period 2 consumers make a decision where to buy the base good. In period 3 consumers observe the price of the add-on and decide whether to buy it or not to buy it.

Period 0:

• The regulator decides whether to educate myopic consumers about the relevance of add-ons.

Period 1:

• Firms choose between uniform pricing and price discrimination.

⁸The demand function $D(x_i)$ is derived from a decision utility of a consumer *a* that is given by $U_{ai} = v_i - q_i + \varepsilon_{ai}$, where v_i represents the quality of a product at firm *i*, q_i represents the price of a product at firm *i*, and ε_{ai} represents a random idiosyncratic preference of consumer *a* that is i.i.d. Depending on the consumer type, the variables v_i and q_i in the decision utility function refer just to the base good or to the base good plus the add-on. It is assumed that all firms produce products with the same quality *v*. See Gabaix and Laibson (2006), p. 532-533, for details on the microfoundation of the demand function $D(x_i)$.

- In case of uniform pricing: Firms pick a price for the base good, p, and a price for the standard add-on, \hat{p} .
- In case of price discrimination: Firms pick a price for the base good, p, a price for the high-priced add-on, \hat{p}_H , and a price for the low-priced add-on, \hat{p}_L .
- Firms decide to make the information about the add-ons shrouded or unshrouded.

Period 2:

- Informed consumers (sophisticated and informed myopic) always take the add-on and its price into consideration for their buying decision. If information about the add-on is *shrouded*, sophisticated consumers and informed myopic consumers form Bayesian posteriors about the unobserved add-on.
- Informed consumers can initiate behavior that allows them to substitute away at costs *e*.
- Uninformed myopic consumers do not consider the add-on for their buying decision.

Period 3:

- Consumers observe the add-on price (if firms have not *unshrouded* prices already).
- All consumers buy the base good.
- Uninformed consumers buy the standard add-on if the price they observe is at most their reservation price \bar{p} .
- Informed consumers buy the standard add-on if the price they observe is at most their substitution costs *e*, or otherwise substitute away.

We formally consider substitution costs only in the period prior to the purchase of the base good (period 2), equivalent to the Gabaix and Laibson (2006) model. In practice, however, substitution costs may occur both in period 2 and in period 3. For example, ... Substitution costs e may reflect costs only for period 2 or joint costs for period 2 and period 3. It has no consequences for the results of our analysis.

2.2 Price equilibria without educating consumers

This sub-section derives symmetric price equilibria of firms without regulatory intervention. It therefore serves as a benchmark for the evaluation of regulatory intervention. As it is shown in the following, the decisions of firms about *shrouding* or *unshrouding* and offering one add-on (uniform pricing) or two add-ons (price discrimination) result in three potential profit maximizing pricing strategies of the firm.

High-priced add-on. This pricing strategy may make sense if the firm can sell the highpriced add-on to a large fraction of uninformed consumers. The firm therefore *shrouds* information about the add-on because *unshrouding* would decrease the fraction of uninformed consumers. In order to maximize profits, the firm sets the price for the add on to $\hat{p} = \bar{p}$, where \bar{p} is the highest possible price for the add-on (i.e., the reservation price of uninformed consumers). Sophisticated consumers who observe that the firm *shrouds* anticipate this, $E\hat{p} = \bar{p}$. The expected profit of a firm is given by

$$\pi_h = (p + \alpha \bar{p}) D(p^* - p). \tag{1}$$

Detailed calculations are in the appendix.

By solving the first-order condition for $p^* = p$, we get a profit maximizing price for the base good of $p = -\alpha \bar{p} + \mu$, where $\mu = \frac{D(0)}{D'(0)}$. It follows that the maximized profit of a firm under this pricing strategy is given by

$$\pi_h^* = \mu D(p^* + \alpha \bar{p} - \mu). \tag{2}$$

Low-priced add-on. This pricing strategy may make sense if the firm can sell the add-on also to sophisticated consumers who substitute away if they observe a price of the add on $\hat{p} > e$. The firm therefore *unshrouds* and sets the price $\hat{p} = e$. All consumers purchase both the base good and the add-on at the firm. The expected profit is given by

$$\pi_l = (p+e)D(p^* - p).$$
(3)

Detailed calculations are in the appendix.

By solving the first-order condition for $p^* = p$, we get a profit maximizing price for the base good of $p = -e + \mu$, where $\mu = \frac{D(0)}{D'(0)}$. It follows that the maximized profit of a firm under this pricing strategy is given by

$$\pi_l^* = \mu D(p^* + e - \mu). \tag{4}$$

Price discrimination. This pricing strategy may make sense if the firm has a technology to separate between myopic and sophisticated consumers, and to make them separate offers. The firm chooses a price for the low-priced add-on, \hat{p}_L , and a price for the high-priced add-on, \hat{p}_H . In order to maximize profits from uninformed consumers, the firm sets the price for the high-priced add on to $\hat{p}_H = \bar{p}$. In order to maximize profits from informed consumers, the firm also offers a low-priced add-on to informed consumers for a price $\hat{p}_L = e$. As discussed in sub-section 2.1, efficiency of price discrimination is limited by *attrition* (β) and *cannibalization* (γ). The expected profit of a firm is given by

$$\pi_d = (p + e((1 - \alpha)\beta + \alpha(1 - \gamma)) + \bar{p}\gamma\alpha)D(-p + p^*)$$
(5)

Detailed calculations are in the appendix.

By solving the first-order condition for $p^* = p$, we get a profit maximizing price for the base good of $p = -e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \mu$, where $\mu = \frac{D(0)}{D'(0)}$. It follows that the maximized profit of a firm under this pricing strategy is given by

$$\pi_d^* = \mu D(p^* + e((1-\alpha)\beta + \alpha(1-\gamma)) + \bar{p}\gamma\alpha - \mu)$$
(6)

Symmetric price equilibria. The firm chooses a pricing strategy to maximize its expected profit. The conditions for a *low-priced add-on* strategy are $\pi_l^* > \max(\pi_h^*, \pi_d^*)$. The conditions for a *high-priced add-on* strategy are $\pi_h^* > \max(\pi_d^*, \pi_l^*)$. The conditions for a *price discrimination* strategy are $\pi_d^* > \max(\pi_h^*, \pi_l^*)$. These conditions imply threshold levels for the fraction of myopic consumers of the population, α , that determine the symmetric price equilibrium of all firms. The threshold level depend on substitution costs e, the upper level for the add-on price \bar{p} , and the efficiency of product discrimination (the attrition effect β , and the cannibalization effect γ).

Proposition 1 (Price equilibria without regulatory intervention). Let

$$\alpha^{\dagger} = \min\left(\frac{e}{\bar{p}}, \frac{e(1-\beta)}{e(1-\beta) + (\bar{p}-e)\gamma}\right)$$
(7)

and

$$\alpha^{\ddagger} = \max\left(\frac{e}{\bar{p}}, \frac{e\beta}{e\beta + (\bar{p} - e)(1 - \gamma)}\right).$$
(8)

The firm chooses a low-priced add-on strategy if $\alpha < \alpha^{\dagger}$, a price discrimination strategy if $\alpha^{\dagger} < \alpha < \alpha^{\ddagger}$, and a high-priced add-on strategy if $\alpha^{\ddagger} < \alpha$.

The proof is in the appendix.⁹

The threshold level that are defined in Proposition 1 are illustrated in Figure 2. The following corollaries represent special cases of Proposition 1.

Corollary 1 (Equilibrium without price discrimination). Price discrimination is not efficient if $\beta \leq 1 - \gamma$, which implies that $a^{\dagger} = \alpha^{\ddagger} = \frac{e}{\bar{p}}$. Then, firms choose a low-priced add-on strategy if $\alpha < \frac{e}{\bar{p}}$ and a high-priced add-on strategy otherwise.

The proof is in the appendix.

Corollary 1 reflects that the positive effect from offering low-priced add-ons to sophisticated and informed myopic consumers, β , is weaker than the cannibalization effect, $1 - \gamma$. Note that the threshold level for α of Corollary 1 is equivalent to the threshold level from Gabaix and Laibson (2006) who analyze price equilibria without price discrimination.

Corollary 2 (Equilibrium with price discrimination and no attrition). Suppose price discrimination involves no attrition of informed consumers ($\beta = 1$). This implies that $\alpha^{\dagger} = 0$ and $\alpha^{\ddagger} = \frac{e}{e + (\bar{p} - e)(1 - \gamma)}$. Then, firms choose a price discrimination strategy if $\alpha < \alpha^{\ddagger}$ and a high-priced add-on strategy otherwise. A low-priced add-on strategy cannot be the optimal strategy.

The proof is in the appendix.

⁹Comment on multiple equilibria. See also Gabaix and Laibson (2006), footnote 30.

Corollary 3 (Equilibrium with price discrimination and no cannibalization). Suppose price discrimination involves no cannibalization of uninformed consumers ($\gamma = 1$). This implies that $\alpha^{\dagger} = \frac{e(1-\beta)}{e(1-\beta)+(\bar{p}-e)}$ and $\alpha^{\ddagger} = 1$. Then, firms choose a low-priced add-on strategy if $\alpha < \alpha^{\dagger}$ and a price discrimination strategy otherwise. A high-priced add-on strategy cannot be the optimal strategy.

The proof is in the appendix.

2.3 Price equilibria with educating consumers

Suppose that the regulator decides to intervene via educating myopic consumers. A fraction λ_R of myopic consumers then becomes *informed consumers* and decides just like sophisticated consumers. Thus, the fraction of uninformed consumers decreases from α to $(1-\lambda_R)\alpha$, and the impact of unshrouding changes from λ_F to $\lambda'_F \in [0, \lambda_F]$. As shown in the previous analysis, λ_F (and equivalently λ'_F) does not impact the optimal pricing strategy of firms.¹⁰

For example, when the optimal strategy of a firm is a low priced add-on for $\alpha < \alpha^{\dagger}$ without regulatory intervention, this strategy is only optimal for $(1 - \lambda_R)\alpha < \alpha^{\dagger} \Leftrightarrow \alpha < \frac{1}{1 - \lambda_R}\alpha^{\dagger}$ with regulatory intervention. The impact on the threshold between a *price discrimination* and a *high priced add-on* pricing strategy is effected equivalently.

Proposition 2 (Price equilibria with regulatory intervention). Suppose the regulator educates a fraction λ_R of myopic consumers about the relevance of the add-on prices, and let

$$\alpha^{\S} = \frac{1}{1 - \lambda_R} \alpha^{\dagger} \tag{9}$$

and

$$\alpha^{\sharp} = \frac{1}{1 - \lambda_R} \alpha^{\ddagger}.$$
 (10)

The firm chooses a low-priced add-on strategy if $\alpha < \alpha^{\S}$, a price discrimination strategy if $\alpha^{\S} < \alpha < \alpha^{\sharp}$, and a high-priced add-on strategy if $\alpha^{\sharp} < \alpha$.

The proof is in the appendix.

¹⁰Note that, however, λ_F or λ'_F impacts the range where two alternative pricing equilibria exist.

The threshold levels that are defined in Proposition 2 are illustrated in Figure 3. Similarly as before (see Corollary 1 to Corollary 3), when $\beta \leq 1 - \gamma$, it follows that price discrimination is not the optimal pricing strategy. Firms choose a *low-priced add-on* strategy if $\alpha < \frac{e}{\bar{p}(1-\lambda)}$ and a *high-priced add-on* strategy otherwise. When $\beta = 1$ which means that price discrimination involves no *attrition* of informed consumers, it follows that a *low-priced add-on* strategy is not the optimal strategy. Firms choose a *price discrimination* strategy if $\alpha < \alpha^{\sharp}$ and a *high-priced add-on* strategy otherwise. When $\gamma = 1$ which means that price discrimination involves no *cannibalization* of uninformed consumers, it follows that a *high-priced add-on* strategy is not the optimal strategy. Firms choose a low-priced add-on strategy if $\alpha < \alpha^{\sharp}$ and a *high-priced add-on* strategy otherwise. When $\gamma = 1$ which means that price discrimination involves no *cannibalization* of uninformed consumers, it follows that a *high-priced add-on* strategy is not the optimal strategy. Firms choose a low-priced add-on strategy if $\alpha < \alpha^{\sharp}$ and a *price discrimination* strategy is not the optimal strategy. Firms choose a low-priced add-on strategy if $\alpha < \alpha^{\$}$ and a *price discrimination* strategy if $\alpha < \alpha^{\$}$ and a *price discrimination* strategy. Firms choose a low-priced add-on strategy if $\alpha < \alpha^{\$}$ and a *price discrimination* strategy.

3 Welfare Effects

This section provides comparative statics for social welfare and consumer protection. First, we discuss generally social welfare and consumer protection issues that arise from different prices for uninformed and informed consumers. Second, we analyze effects of price discrimination. This is important for potential actions of the regulator. Third, we analyze effects of educating consumers and derive conditions when it is beneficial for social welfare and when it has adverse effects. Fourth, we discuss the case for subsidized public firms.

3.1 Social welfare and consumer protection

As suggested by Camerer, Issacharoff, Loewenstein, O'Donoghue, and Rabin (2003) in an influential article on *asymmetric paternalism*, regulation should "create large benefits for those who make errors, while imposing little or no harm on those who are fully rational" (p. 1212). Furthermore, regulation should "increase economic efficiency as measured by the sum of consumer and producer surplus" (p. 1220). Our welfare analysis follows this line of thought.

The measure for *social welfare loss* is given by the number of consumers who substitute away from the add-on multiplied by their effort costs e.

Besides social welfare, the pricing strategy of firms also affects consumer protection. From an equality perspective, one may want to prevent that informed consumers benefit from the high add-on prices paid by uninformed consumers. However, it is not so clear if regulation and consumer policy is the right tool to create equality. It is often argued that not competition policy but other instruments such as taxation should be used for income redistribution between consumers (e.g., Motta, 2004; Armstrong, 2008). Furthermore, consumers should have the incentive to become informed, because competition between firms may otherwise be weakened, which leads to reduced consumer welfare (Armstrong, Vickers, and Zhou, 2009).

In any case, several alternative measure could be used to evaluate consumer protection. A simple measure for consumer protection is the fraction of exploited consumers. The fraction of exploited consumers obviously decreases through education whenever a *high-priced add-on* equilibrium or a *price discrimination* equilibrium exists.

Expected total prices of myopic consumers reflect that a myopic consumer may become informed but may also stay uninformed after education. The measure is calculated as total prices of an uninformed consumer times the probability that a myopic consumer stays uninformed $(1 - \lambda_R)$, plus total prices of an informed consumer times the probability that a myopic consumer becomes informed (λ_R) .

3.2 Welfare effects of price discrimination

Proposition 3 (price discrimination and social welfare). Price discrimination decreases social welfare under an otherwise low-priced add-on equilibrium ($\alpha^{\dagger} < \alpha < \frac{e}{\bar{p}}$). It increases social welfare under an otherwise high-priced add-on equilibrium ($\frac{e}{\bar{p}} < \alpha < \alpha^{\ddagger}$).

Figure 4 illustrates effects of price discrimination on social welfare. Besides social welfare, the pricing strategy of firms also affects cross-subsidies from uninformed (myopic) to informed (sophisticated and informed myopic) consumers.

Proposition 4 (price discrimination and prices). Price discrimination always decreases prices of sophisticated consumers ($\alpha^{\dagger} < \alpha < \alpha^{\ddagger}$). The effect of price discrimination on prices of myopic consumers depends on the add-on that myopic consumers are offered. For myopic consumers who are offered the low-priced add-ons, it always decreases prices $(\alpha^{\dagger} < \alpha < \alpha^{\ddagger})$. For myopic consumers who are offered the high-priced add-ons, it decreases prices under an otherwise high-priced add-on equilibrium $(\frac{e}{\bar{p}} < \alpha < \alpha^{\ddagger})$ and increases prices under an otherwise low-priced add-on equilibrium $(\alpha^{\dagger} < \alpha < \frac{e}{\bar{p}})$. Similarly, it decreases expected prices of myopic consumers under an otherwise high-priced add-on equilibrium $(\alpha^{\dagger} < \alpha < \frac{e}{\bar{p}})$. Similarly, it decreases expected prices of myopic consumers under an otherwise high-priced add-on equilibrium $(\alpha^{\dagger} < \alpha < \frac{e}{\bar{p}})$.

Figure 5 illustrates effects of price discrimination on prices of uninformed and informed consumers.

3.3 Welfare effects of educating consumers

Proposition 5 (education and social welfare). Educating consumers increases social welfare, if it changes a high-priced add-on equilibrium ($\alpha^{\ddagger} < \alpha < \alpha^{\ddagger}$) or a price discrimination equilibrium ($\alpha^{\ddagger} < \alpha < \alpha^{\$}$). It decreases social welfare, if a high-priced add-on equilibrium ($\alpha^{\ddagger} < \alpha$) or a price discrimination equilibrium ($\alpha^{\$} < \alpha < \alpha^{\ddagger}$) remains.

The proof is in the appendix.

Table 1 shows the welfare loss or welfare gain from educating consumers for each range of α . Figure 6 illustrates effects of educating consumers on social welfare.

So far, the analysis has focussed on varying fractions of myopic consumers α . The efficiency of educating myopic consumers that is reflected in λ is also important, especially, if the regulator can increase λ with high education efforts or with the right strategy. Ideally, the regulator makes enough myopic consumers informed such that firms choose the socially optimal *low-priced add-on* equilibrium. However, it is often not possible for the regulator to make a sufficiently large fraction of myopic consumers informed for at least two reasons: first, experimental studies show that a surprisingly low fraction of consumers changes its behavior based on new important information (e.g., Choi, Laibson, and Madrian, 2010), and second, firms may add complexity to their price structures in order to prevent that consumers become informed (Carlin, 2009). A special case where price discrimination is not efficient ($\beta \leq 1 - \gamma$) and only a *low-priced* add-on or a high-priced add-on equilibrium exists, is illustrated in Figure 7. A special case where price discrimination is efficient without attrition of informed consumers ($\beta = 1$) and only a price discrimination or a high-priced add-on equilibrium exists, is illustrated in Figure 8. These figures show as well consequences for a varying fraction of myopic consumers α as consequences for a varying effect of educating myopic consumers λ .

Figure 9 illustrates how education affects total prices (base good and add-on) of uninformed and informed consumers. It shows that total prices for uninformed consumers increase for a lower α , until a *low-priced add-on* equilibrium is reached. Total prices for informed consumers also increase for a lower α until a *low-priced add-on* equilibrium is reached. The *expected* total prices of myopic consumers with education are largely below total prices without education (for $\alpha < 0.7$), even within a *high-priced add-on* or *price discrimination* equilibrium. Interestingly, for a high fraction of myopic consumers ($\alpha > 0.7$), education even increases *expected* total prices of myopic consumers.

Proposition 6 (education and expected prices). Let

$$\alpha^{\diamondsuit} = \max(1 - \frac{e}{\bar{p}}, \alpha^{\sharp}). \tag{11}$$

Educating consumers increases expected prices for myopic consumers for $\alpha^{\diamondsuit} < \alpha$. It decreases expected prices for $\alpha^{\dagger} < \alpha < \alpha^{\diamondsuit}$.

The proof is in the appendix.

Pitfalls for the regulator As shown in the previous analysis, effects of educating myopic consumers depend on price equilibria, which in turn depend on the pricing flexibility of firms. The regulator can evaluate effects of educating myopic consumers if full information about the pricing dynamics is available. In many situations, however, this will not be the case.

Figure 10 illustrates a situation when effects of educating consumers depend on the pricing flexibility of firms. Suppose the regulator observes a high-priced add-on equilibrium. The regulator estimates that a fraction $\alpha = 0.6$ of consumers is myopic, and that education could

make a fraction $\lambda = 0.3$ of myopic consumers informed. Without knowing if firms own or could acquire a technology for price discrimination, it's not possible for the regulator to evaluate effects of educating consumers. If firms can only apply uniform pricing strategies, educating consumers decreases social welfare. If firms own a technology for price discrimination, educating consumers increases social welfare.

4 Alternative ways of regulatory intervention

This section discusses how the regulator can impact pricing strategies, consumer behavior and social welfare through subsidized public firms when an equilibrium pricing strategy exists where firms offer high-priced-add-ons to uninformed consumers.

A "public firm" represents a firm that accepts to follow a certain (loss-making) pricing strategy suggested by the regulator, and in turn receives a subsidy from the regulator. Subsidizing a public firm is different from a pure redistribution of income from taxpayers to uninformed consumers because it directly affects the purchase decision of consumers and the related social welfare loss. Thus, it may be a better way for creating equality than the redistribution of income through taxes or other isolated instruments. For the evaluation of such an intervention, costs for subsidizing a public firm are compared to a potential social welfare gain which reflects the benefit of all consumers from lower prices net taxes to finance subsidies.

The benefit of regulatory intervention is that social welfare increases for every informed consumer who does not substitute away, but would have substituted away without the public firm. The drawback of such a strategy is that it may require high subsidies for the public firm. Although the social welfare gain is net of subsidies, a strategy where huge subsidies are necessary in order to create a relatively small net social welfare gain does not seem to be justified. The measure we use to evaluate the efficiency of subsidies to a public firm is the costs of a subsidy divided by the social welfare gain, which is the lower the better.

4.1 Subsidizing add-ons

Consider a strategy where the regulator pays subsidies to consumers for buying certain products. E.g., consumers get tax breaks for investing in retirement funds. The rationale behind such a strategy may be the observation that a fraction of consumers does not buy a product that is socially desirable, a tax break makes the product more attractive, and consequently, more consumers buy it.

However, when the subsidized product is a high-priced add-on that myopic consumers already buy and sophisticated consumers avoid, it is easy to see that subsidies have purely adverse effects. Subsidies increase the reservation price \bar{p} , which makes a high-priced add-on strategy more profitable. Consequently, the thresholds for α where a low-priced add-on equilibrium or a price discrimination equilibrium exist decrease (see also Proposition 1 and Proposition 2). Thus, when subsidies for add-ons lead to a high-priced add-on equilibrium instead of a low-priced add-on or price discrimination equilibrium, or to a price discrimination add-on equilibrium instead of a low-priced add-on equilibrium, direct consequences of subsidies are a higher social welfare loss.

When subsidies do not change a price equilibrium, there is no effect on social welfare. However, it has redistribution effects among consumers. Profits of firms are unchanged because the higher prices on the add-ons get competed away via the base good. The price of the add-on increases from \bar{p} to $\bar{p} + \tau$, where τ represents the value of the tax break. The price of the base good decreases from $-\alpha \bar{p} + \mu$ to $-\alpha(\bar{p} + \tau) + \mu$ in a high-priced add-on equilibrium. In sum, uninformed consumers who buy the base good and the add-on pay $(1 - \alpha)\tau$ less and informed consumers who only buy the base good pay $\alpha\tau$ less. Thus, uninformed consumers profit relatively more for a low α , and informed consumers profit relatively more for a high α . Subsidies, of course, have to be financed by all consumers via higher taxes. Consequences are equivalent under a price discrimination equilibrium. In sum, when subsidies for add-ons do not change a prices equilibrium, which is the best possible case, it has no effect on social welfare but leads to an obscure redistribution of income among consumers.

4.2 Offering low-priced add-ons to all consumers

Suppose that the suggested strategy is to offer the base good for the equilibrium market price, i.e., $p = -\alpha \bar{p} + \mu$, and the add-on for *e*. Recall that such a pricing strategy cannot exist for *for-profit* firms, as shown in the previous analysis. The following analysis assumes that the total volume of potential subsidies is such that all firms can participate in the subsidized pricing strategy.

A detailed description of this strategy is provided in Figure 12. Consider a high-priced add-on equilibrium without education, and suppose price discrimination is not attractive, i.e., the threshold between a low-priced add-on and a high-priced add-on equilibrium is $\alpha^{\dagger} = \alpha^{\ddagger} = \frac{e}{\bar{p}}$ (see Corollary 1). Then, subsidies are given by $\alpha \bar{p} - e$. For $\alpha \to \alpha^{\ddagger} = \frac{e}{\bar{p}}$ (from above), subsidies $S_h \to 0$ and the social welfare gain $G_h \to e(1 - \frac{e}{\bar{p}})$. Thus, this strategy is very attractive for relatively low α . For $\alpha \to 1$, subsidies are $S_h \to \bar{p} - e$ and the social welfare gain $G_h \to 0$. This is obviously unattractive. The intuition is the following: The firm could sell the add-on for a price \bar{p} to all consumers under a high-priced add-on equilibrium. When the firm offers the add-on for e instead of \bar{p} to all consumers, the firm looses some of its profit with uninformed consumers. The lower the fraction of uninformed consumers, they get an additional profit from informed consumers who otherwise substitute away, i.e., $e(1 - \alpha)$. The higher the fraction of informed consumers, i.e., the lower α , the higher this additional profit. Thus, required subsidies to compensate firms for a loss-making price strategy are increasing in α .

Proposition 7 (offering of low-priced add-on for all consumers). Suppose the regulator subsidizes firms that offer uninformed consumers low-priced add-ons, and does not educate consumers. The efficiency of this intervention, defined as the cost of the required subsidy divided by the net social welfare gain, is

$$E_h = \frac{\alpha \bar{p} - e}{e(1 - \alpha)} \tag{12}$$

under a high-priced add-on equilibrium ($\alpha^{\ddagger} < \alpha$).

The proof is in the appendix.

Proposition 7 yields an important result: If a public firm makes losses, the firm is not necessarily inefficient. In fact, the firm may be very efficient in redistributing income to uninformed consumers.

Recall that E_h is defined such that the lower the better. This measure increases in α . Therefore, educating consumers (lowering the fraction of uninformed consumers) increases the efficiency of this regulatory intervention.

4.3 Offering low-priced add-ons to informed consumers only

From the perspective of a "paternalistic" regulator who cares for myopic consumers, a strategy looks attractive where a public firm offers products to myopic consumers while private firms offer products to sophisticated consumers. Then, private firms would choose a *low-priced add-on* pricing strategy, and there would be no social welfare loss or consumer protection issue. However, when the public firm offers the base good at a price that is attractive to myopic consumers, this offer is also attractive to sophisticated consumers. Therefore, a public firm cannot increase its market share only with myopic consumers.

However, offers that only attract informed consumers are feasible. For such a strategy, the net anticipated surplus of informed consumers, i.e., $\hat{x}_i = -p - \min\{E\hat{p}, e\} + p^* + \min\{E\hat{p}^*, e\}$, needs to be higher at the public firm than at the private firm, while the net anticipated surplus of uninformed consumers, i.e., $\bar{x}_i = -p + p^*$, needs to be lower at the public firm. Therefore, suppose that the strategy is to offer the base good above the equilibrium market price, i.e., $p + \Delta$, and the add-on for $\hat{e} - \Delta$, where $\hat{e} < e$.

The intuition for this strategy is as follows: When a private firm only attracts myopic consumers, this leads to the so-called *loss leader pricing*. The price of the base good is $p = -\bar{p} + \mu$ and the price of the add-on is $\hat{p} = \bar{p}$. Myopic consumers do NOT subsidize sophisticated consumers and are NOT exploited. At first glance, private firms may be happy that they don't have to serve informed consumers. However, this benefit gets competed away if all private firms enjoy this effect, which results in a *loss leader pricing*. The public firm attracts informed consumers who do not need to substitute away. In a high-priced add-on equilibrium, social welfare gain is $(1 - \alpha)e$.

A detailed description of this strategy is provided in Figure 13. Similarly as in Section 4.2, consider a high-priced add-on equilibrium without education, and suppose price discrimination is not attractive, i.e., $\alpha^{\dagger} = \alpha^{\ddagger} = \frac{e}{\bar{p}}$. Then, subsidies are given by $(\bar{p}-\hat{e})(1-\alpha)$. For $\alpha \to \alpha^{\ddagger} = \frac{e}{\bar{p}}$ (from above), subsidies $S_h \to \bar{p} - 2e + \frac{e^2}{\bar{p}}$. Subsidies are decreasing in α . For $\alpha = 1$, subsidies are 0. The intuition is the following: the market price of the base good decrease by $\bar{p}(1-\alpha)$ through the intervention. Thus, the lower α , the higher the loss of firms on the base good. Moreover, on the positive side, firms get additional profits of $e(1-\alpha)$ from informed consumers. The lower α , the lower the additional profits. In sum, the lower α , the higher the required subsidies for firms in order to compensate them for losses if they accept the suggested pricing strategy of the regulator.

Proposition 8 (offering of low-priced add-ons for informed consumers). Suppose the regulator subsidizes firms that offer informed consumers low-priced add-ons, and does not educate consumers. The efficiency of this intervention, defined as the cost of the required subsidy divided by the net social welfare gain, is

$$E_h = \frac{\bar{p} - \hat{e}}{e} \tag{13}$$

under a high-priced add-on equilibrium ($\alpha^{\ddagger} < \alpha$).

The proof is in the appendix.

The measure is constant in α . Therefore, educating consumers does not matter for the efficiency of this regulatory strategy, and the goal should be to minimize the total volume of subsidies. For this strategy this is $S_h = (\bar{p} - \hat{e})(1 - \alpha)$. Subsidies decrease in α . Therefore, educating consumers (lowering the fraction of uninformed consumers) increases subsidies, and, unless it changes the equilibrium price strategy of firms, should not be carried out.

4.4 Educating consumers and optimal subsidization

Including regulatory intervention through educating consumers, we presented four alternative regulatory strategies: first, educating consumers, and second, directly subsidizing add-ons,

third, subsidizing firms that offer low-priced add-ons to all consumers, and fourth, subsidizing firms that offer low-priced add-ons to informed consumers. In this section, we discuss the optimal regulatory strategy. The rationale for the evaluation of alternative intervention strategies is that the regulator should use the strategy that first, yields the highest social welfare gain, and second, requires the least subsidies (taxes).

It is easy to see that whenever the regulator can achieve a *low-priced add-on* equilibrium through educating consumers, this is the optimal strategy and other interventions are not necessary. When the regulator can change a *high-priced add-on* equilibrium to a *price discrimination* equilibrium through educating consumers, this strategy should be combined with subsidizing public firms. When the regulator cannot change a *high-priced add-on* equilibrium or a *price discrimination* equilibrium through educating consumers, educating consumers may be beneficial or counterproductive, depending on the details how firms are subsidized.

When considering to subsidize firms, the regulator can choose between two reasonable strategies: subsidizing firms that offer low-priced add-ons to all consumers, as discussed in Section 4.2, or subsidizing firms that offer low-priced add-ons only only to informed consumers, as discussed in Section 4.3. Subsidizing firms that offer low-priced add-ons only to uninformed consumers is not feasible, and directly subsidizing add-ons is never optimal.

With both relevant strategies, the regulator can avoid any social welfare loss that would exist under a *high-priced add-on* equilibrium. Thus, the strategies only differ in the amount of required subsidies, and the regulator should choose the strategy where the subsidies are minimized.

Proposition 9 (subsidizing firms). The regulator can increase social welfare through subsidizing firms that accept loss-making price strategies when a high-priced add-on equilibrium exists. Let

$$\alpha^{H} = \frac{\bar{p} - \hat{e} + e}{\bar{p} - \hat{e} + \bar{p}}.$$
(14)

The optimal strategy that requires the lowest subsidies for the same net effect on social welfare is subsidizing firms that offer low-priced add-ons to all consumers for $\alpha < \alpha^H$ and, otherwise, subsidizing firms that offer low-priced add-ons only to informed consumers. See Figure 14 for an illustration.

5 Conclusion

The goal of educating consumers is that firms offer the base good and the add-on at "fair" prices that are attractive for all consumers. If educating consumers changes a high-priced add-on equilibrium or a price discrimination equilibrium of firms, social welfare increases. However, educating consumers is not always successful in reaching this goal because a fraction of consumers may not be receptive for new information or not be able to use any but the most simple decision rationale. Then, as we show in this paper, educating consumers may decrease social welfare. Effects depend on the fraction of myopic consumers, the efficiency of educating myopic consumers, the effort costs of substituting away from add-ons, the upper level for the add-on price, and the efficiency of price discrimination. When educating consumers does not lead to a new price equilibrium where firms offer low-priced add-ons, the regulator should intervene in the product market. We define a "public firm" as a firm that accepts to follow a certain (loss-making) pricing strategy suggested by the regulator, and in turn receives a subsidy from the regulator. We develop alternative regulatory strategies with different comparative advantages that each create a net social welfare gain. Depending on the specific situation, the optimal regulatory strategy may be 1) only educating consumer, 2) educating consumers and subsidizing firms, or 3) only subsidizing firms.

Our paper develops a framework to analyze effects of alternative regulatory interventions and to derive the optimal regulatory strategy. In summary, the regulator needs to carefully analyze the situation before intervening via educating myopic consumers and may find no intervention at all or other regulatory strategies more beneficial.

References

Armstrong, M., 2006, Recent developments in the economics of price discrimination, in Richard Blundell, Whitney K. Newey, and Torsten Persson, ed.: Advances in Economics and Econometrics: Theory and Applications: Ninth World Congress, vol. 2 pp. 97–141. Cambridge University Press.

- , J. Vickers, and J. Zhou, 2009, Consumer protection and the incentive to become informed, Journal of the European Economic Association 7, 399–410.
- Ausubel, L.M., 1991, The failure of competition in the credit card market, The American Economic Review 81, 50–81.
- Beshears, J., J. J. Choi, D. Laibson, and B. Madrian, 2009, How Does Simplified Disclosure Affect Individuals' Mutual Fund Choices?, *NBER Working Paper*.
- Camerer, C., S. Issacharoff, G. Loewenstein, T. O'Donoghue, and M. Rabin, 2003, Regulation for Conservatives: Behavioral Economics and the Case for" Asymmetric Paternalism", University of Pennsylvania Law Review 151, 1211–1254.
- Carlin, B.I., 2009, Strategic price complexity in retail financial markets, Journal of Financial Economics 91, 278 – 287.
- Choi, J., D. Laibson, and B.C. Madrian, 2010, Why does the law of one price fail? An experiment on index mutual funds, *Review of Financial Studies* 23, 1405–1432.
- De Meza, D., B. Irlenbusch, and D. Reyniers, 2008, Financial capability: A behavioural economics perspective, *Consumer Research* 69.
- Esteves, R.B., 2009, A survey on the economics of behaviour-based price discrimination, Nipe working papers NIPE Universidade do Minho.
- Fudenberg, D., and J.M. Villas-Boas, 2006, Behavior-based price discrimination and customer recognition, in T. Hendershott, ed.: *Handbooks in Information Systems*, vol. 1 . chap. 7, pp. 377–436 (Elsevier B.V.).
- Gabaix, X., and D. Laibson, 2006, Shrouded attributes, consumer myopia, and information suppression in competitive markets, *The Quarterly Journal of Economics* 121, 505–540.

- Jolls, Ch., and C.R. Sunstein, 2006, Debiasing through law, *Journal of Legal Studies* 35, 199–241.
- Lal, R., and C. Matutes, 1994, Retail pricing and advertising strategies, Journal of Business 67, 345–370.
- Motta, M., 2004, Competition policy: theory and practice (Cambridge University Press).
- Shapiro, C., 1994, Aftermarkets and consumer welfare: Making sense of Kodak, Antitrust Law Journal 63, 483–511.
- ———, and H.R. Varian, 1998, Versioning: the smart way to sell information, *Harvard Business Review* November/December, 106–114.
- Spiegler, R., 2006a, Competition over agents with boundedly rational expectations, *Theoret*ical Economics 1, 207–231.
- ———, 2006b, The market for quacks, *Review of Economic Studies* 73, 1113–1131.
- Stole, L.A., 2007, Price discrimination and competition, in Mark Armstrong, and Robert Porter, ed.: Handbook of Industrial Organization, vol. 3 . pp. 2221–2300 (Elsevier).

A Appendix: Calculations and proofs

A.1 Calculations and proofs for Section 2.2

Equation 1 (high-priced add-on w/o regulatory intervention). The expected profit of a firm is composed of the expected profit from sales to informed consumers who only purchase the base good, $p(1 - \alpha)D(\hat{x}_i)$, and the expected profit from sales to uninformed consumers who buy both the base good and the add-on, $(p + \hat{p})\alpha D(\bar{x}_i)$. Thus, the expected profit is given by

$$\pi_{h} = p(1-\alpha)D(\hat{x}_{i}) + (p+\hat{p})\alpha D(\bar{x}_{i})$$

$$= p(1-\alpha)D(-p - \min\{E\hat{p}, e\} + p^{*} + \min\{E\hat{p}^{*}, e\})$$

$$+ (p+\hat{p})\alpha D(-p + p^{*})$$

$$= p(1-\alpha)D(-p - e + p^{*} + e) + (p+\bar{p})\alpha D(-p + p^{*})$$

$$= (p+\alpha\bar{p})D(p^{*} - p)$$

Equation 3 (low-priced add-on w/o regulatory intervention). The expected profit of a firm is composed of the expected profit from sales to informed (sophisticated and informed myopic) consumers, $(p + \hat{p})(1 - (1 - \lambda_F)\alpha)D(\hat{x}_i)$, and the expected profit from sales to uninformed myopic consumers, $(p + \hat{p})(1 - \lambda_F)\alpha D(\bar{x}_i)$. All consumers purchase both the base good and the add-on at the firm. Thus, the expected profit is given by

$$\pi_{l} = (p+\hat{p})(1-(1-\lambda_{F})\alpha)D(-p-\min\{E\hat{p},e\}+p^{*}+\min\{E\hat{p}^{*},e\})$$
$$+(p+\hat{p})(1-\lambda_{F})\alpha D(-p+p^{*})$$
$$= (p+e)(1-(1-\lambda_{F})\alpha)D(-p-e+p^{*}+e)+(p+e)(1-\lambda_{F})\alpha D(-p+p^{*})$$
$$= (p+e)D(p^{*}-p)$$

Equation 5 (price discrimination w/o regulatory intervention). Recall that price discrimination may not be fully efficient. First, because of *attrition*, only a fraction β of informed consumers (pre advertising) buys it at the firm, and a fraction $1-\beta$ substitutes away.

Second, because of product cannibalization, a fraction $1 - \gamma$ of myopic consumers becomes informed, and a fraction γ stays uninformed. The expected profit of a firm is composed of the expected profit from sales to informed consumers who substitute away, $p(1-\alpha)(1-\beta)D(\hat{x})$, sales to informed consumers who buy the low-priced add-on, $(p + e)(1 - \alpha)\beta D(\hat{x})$, sales to informed consumers (myopic consumers who got informed through product cannibalization) who buy the low-priced add-on, $(p + e)(1 - \lambda_F)\alpha(1 - \gamma)D(\hat{x})$, sales to uninformed consumers (myopic consumers who don't get informed but get the low-priced add-on offered) who buy the low-priced add-on, $(p + e)\lambda_F\alpha(1 - \gamma)D(\bar{x})$, and sales to uninformed myopic consumers who buy the high-priced add-on, $(p + \bar{p})\alpha\gamma D(\bar{x})$. Thus, the expected profit of a firm is given by

$$\begin{aligned} \pi_d &= p(1-\alpha)(1-\beta)D(\hat{x}) + (p+e)(1-\alpha)\beta D(\hat{x}) \\ &+ (p+e)(1-\lambda_F)\alpha(1-\gamma)D(\hat{x}) + (p+e)\lambda_F\alpha(1-\gamma)D(\bar{x}) \\ &+ (p+\bar{p})\alpha\gamma D(\bar{x}) \end{aligned} \\ &= p(1-\alpha)(1-\beta)D(-p+p^*) + (p+e)(1-\alpha)\beta D(-p+p^*) \\ &+ (p+e)(1-\lambda_F)\alpha(1-\gamma)D(-p+p^*) + (p+e)\lambda_F\alpha(1-\gamma)D(-p+p^*) \\ &+ (p+\bar{p})\alpha\gamma D(-p+p^*) \end{aligned} \\ &= [p(1-\alpha)(1-\beta) + (p+e)(1-\alpha)\beta + (p+e)\alpha(1-\gamma) + (p+\bar{p})\alpha\gamma] D(-p+p^*) \\ &= [p+e((1-\alpha)\beta + \alpha(1-\gamma)) + \bar{p}\gamma\alpha] D(-p+p^*) \end{aligned}$$

Proposition 1 (price equilibria w/o regulatory intervention). The firm chooses a pricing strategy to maximize its expected profit. The conditions for a *low-priced add-on* strategy are that $\pi_l^* > \pi_h^*$ and $\pi_l^* > \pi_d^*$, which is equivalent to $\pi_l^* > \max(\pi_h^*, \pi_d^*)$. First, we get

$$\pi_l^* > \pi_h^*$$

$$\mu D(p^* + e - \mu) > \mu D(p^* + \alpha \bar{p} - \mu)$$

$$e > \alpha \bar{p}$$

$$\frac{e}{\bar{p}} > \alpha.$$

Second, we get

$$\begin{aligned} \pi_l^* &> \pi_d^* \\ \mu D(p^* + e - \mu) &> \mu D(p^* + e((1 - \alpha)\beta + \alpha(1 - \gamma)) + \bar{p}\gamma\alpha - \mu)) \\ e &> e((1 - \alpha)\beta + \alpha(1 - \gamma)) + \bar{p}\gamma\alpha \\ e &> e\beta - e\alpha\beta + e\alpha(1 - \gamma) + \bar{p}\gamma\alpha \\ e - e\beta &> \alpha(-e\beta + e - e\gamma + \bar{p}\gamma) \\ e - e\beta &> \alpha(e(1 - \beta) + (\bar{p} - e)\gamma) \\ \frac{e(1 - \beta)}{e(1 - \beta) + (\bar{p} - e)\gamma} &> \alpha. \end{aligned}$$

Thus, the conditions for a *low-priced add-on* strategy hold if $\alpha < \frac{e}{\bar{p}}$ and $\alpha < \frac{e(1-\beta)}{e(1-\beta)+(\bar{p}-e)\gamma}$, which can be written as $\alpha < \alpha^{\dagger}$, where

$$\alpha^{\dagger} = \min\left(\frac{e}{\bar{p}}, \frac{e(1-\beta)}{e(1-\beta) + (\bar{p}-e)\gamma}\right).$$

The conditions for a high-priced add-on strategy are $\pi_h^* > \pi_d^*$ and $\pi_h^* > \pi_l^*$, which is equivalent to $\pi_h^* > \max(\pi_d^*, \pi_l^*)$. First, we get

$$\begin{split} \pi_h^* &> \pi_d^* \\ \mu D(p^* + \alpha \bar{p} - \mu) &> \mu D(p^* + e((1 - \alpha)\beta + \alpha(1 - \gamma)) + \bar{p}\gamma\alpha - \mu)) \\ \alpha \bar{p} &> e((1 - \alpha)\beta + \alpha(1 - \gamma)) + \bar{p}\gamma\alpha \\ \alpha \bar{p} &> e\beta - e\alpha\beta + e\alpha(1 - \gamma) + \bar{p}\gamma\alpha \\ \alpha \bar{p} &> e\beta - \alpha(e\beta - e + e\gamma - \bar{p}\gamma) \\ \alpha \bar{p} + \alpha(e\beta - e + e\gamma - \bar{p}\gamma) &> e\beta \\ \alpha(\bar{p} + e\beta - e + e\gamma - \bar{p}\gamma) &> e\beta \\ \alpha(e\beta + \bar{p}(1 - \gamma) - e(1 - \gamma)) &> e\beta \\ \alpha(e\beta + (\bar{p} - e)(1 - \gamma)) &> e\beta \\ \alpha &> \frac{e\beta}{e\beta + (\bar{p} - e)(1 - \gamma)}. \end{split}$$

Second, we get

$$\pi_h^* > \pi_l^*$$

$$\mu D(p^* + \alpha \bar{p} - \mu) > \mu D(p^* + e - \mu)$$

$$\alpha \bar{p} > e$$

$$\alpha > \frac{e}{\bar{p}}.$$

Thus, the conditions for a high-priced add-on strategy hold if $\alpha > \frac{e\beta}{e\beta + (\bar{p}-e)(1-\gamma)}$ and $\alpha > \frac{e}{\bar{p}}$, which can be written as $\alpha > \alpha^{\ddagger}$, where

$$\alpha^{\ddagger} = \max\left(\frac{e}{\bar{p}}, \frac{e\beta}{e\beta + (\bar{p} - e)(1 - \gamma)}\right).$$

The conditions for a *price discrimination* strategy are $\pi_d^* > \pi_h^*$ and $\pi_d^* > \pi_l^*$, which is equivalent to $\pi_d^* > \max(\pi_h^*, \pi_l^*)$. From the above calculations it is easy to see that these conditions hold if $\alpha^{\dagger} < \alpha < \alpha^{\ddagger}$. Q.E.D.

Multiple equilibria. If any firm unshrouds, which changes the fraction of myopic consumers from α to $(1 - \lambda_F)\alpha$, the expected profit of a high-priced add-on strategy decreases to

$$\pi'_{h} = p(1 - (1 - \lambda_{F})\alpha)D(-p - \min\{E\hat{p}, e\} + p^{*} + \min\{E\hat{p}^{*}, e\})$$

+(p + \hat{p})(1 - λ_{F}) $\alpha D(-p + p^{*})$
= $p(1 - (1 - \lambda_{F})\alpha)D(-p - e + p^{*} + e) + (p + \bar{p})(1 - \lambda_{F})\alpha D(-p + p^{*})$
= $(p + (1 - \lambda_{F})\alpha\bar{p})D(p^{*} - p)$

For the optimal pricing strategy of firms (see Proposition 1) if follows that more than one equilibrium price strategy may exist.

Corollary 1 (equilibrium w/o price discrimination). The threshold level α^{\dagger} and α^{\ddagger} are defined as

$$\alpha^{\dagger} = \min\left(\frac{e}{\bar{p}}, \frac{e(1-\beta)}{e(1-\beta) + (\bar{p}-e)\gamma}\right)$$

and

$$\alpha^{\ddagger} = \max\left(\frac{e}{\bar{p}}, \frac{e\beta}{e\beta + (\bar{p} - e)(1 - \gamma)}\right).$$

These threshold level imply that $\alpha^{\dagger} \leq \frac{e}{\bar{p}}$ and $\alpha^{\ddagger} \geq \frac{e}{\bar{p}}$. It follows that $\alpha^{\dagger} = \alpha^{\ddagger}$ if, and only if, $\alpha^{\dagger} = \frac{e}{\bar{p}}$ and $\alpha^{\ddagger} = \frac{e}{\bar{p}}$. First, $\alpha^{\dagger} = \frac{e}{\bar{p}}$, if $\frac{e}{\bar{p}} \leq \frac{e(1-\beta)}{e(1-\beta)+(\bar{p}-e)\gamma}$. This is the case if $\beta \leq 1-\gamma$. Second, $\alpha^{\ddagger} = \frac{e}{\bar{p}}$, if $\frac{e}{\bar{p}} \geq \frac{e\beta}{e\beta+(\bar{p}-e)(1-\gamma)}$. This is the case if $\beta \leq \frac{\bar{p}-e}{e\bar{p}-e}(1-\gamma)$. Based on the assumptions that $\bar{p} \in (0,1]$ and $e < \bar{p}$, we know that $1 - \gamma < \frac{\bar{p}-e}{e\bar{p}-e}(1-\gamma)$. It follows that $\alpha^{\dagger} = \alpha^{\ddagger} = \frac{e}{\bar{p}}$, if $\beta \leq (1-\gamma)$. Q.E.D.

Corollary 2 (equilibrium with price discrimination and no attrition). The corollary follows directly from using $\beta = 1$ and from Proposition 1, where

$$\alpha^{\dagger} = \min\left(\frac{e}{\bar{p}}, \frac{e(1-\beta)}{e(1-\beta) + (\bar{p}-e)\gamma}\right)$$
(15)

and

$$\alpha^{\ddagger} = \max\left(\frac{e}{\bar{p}}, \frac{e\beta}{e\beta + (\bar{p} - e)(1 - \gamma)}\right).$$
(16)

Then, $\alpha^{\dagger} = 0$ and $\alpha^{\ddagger} = \frac{e}{e + (\bar{p} - e)(1 - \gamma)}$. The statement of Proposition 1 becomes: The firm chooses a *low-priced add-on* strategy if $\alpha < 0$, a *price discrimination* strategy if $0 < \alpha < \frac{e}{e + (\bar{p} - e)(1 - \gamma)}$, and a *high-priced add-on* strategy if $\frac{e}{e + (\bar{p} - e)(1 - \gamma)} < \alpha$. Given that $\alpha \in [0, 1]$, this is equivalent with the stated results. Q.E.D.

Corollary 3 (equilibrium with price discrimination and no cannibalization). The corollary follows directly and from using $\gamma = 1$ and from Proposition 1, where

$$\alpha^{\dagger} = \min\left(\frac{e}{\bar{p}}, \frac{e(1-\beta)}{e(1-\beta) + (\bar{p}-e)\gamma}\right)$$
(17)

and

$$\alpha^{\ddagger} = \max\left(\frac{e}{\bar{p}}, \frac{e\beta}{e\beta + (\bar{p} - e)(1 - \gamma)}\right).$$
(18)

Then, $\alpha^{\dagger} = \frac{e(1-\beta)}{e(1-\beta)+(\bar{p}-e)}$ and $\alpha^{\ddagger} = 1$. The statement of Proposition 1 becomes: The firm chooses a *low-priced add-on* strategy if $\alpha < \frac{e(1-\beta)}{e(1-\beta)+(\bar{p}-e)}$, a *price discrimination* strategy if $\frac{e(1-\beta)}{e(1-\beta)+(\bar{p}-e)} < \alpha < 1$, and a *high-priced add-on* strategy if $1 < \alpha$. Given that $\alpha \in [0, 1]$, this is equivalent with the stated results. Q.E.D.

A.2 Calculations and proofs for Section 2.3

Proposition 2 (price equilibria with regulatory intervention). Regulatory intervention through educating consumers changes the fraction of informed consumers that firms consider for their pricing strategies from α to $(1 - \lambda_R)\alpha$. Using the results from Proposition 1, it follows that the firm chooses a *low-priced add-on* strategy if $(1 - \lambda_R)\alpha < \alpha^{\dagger}$, a price discrimination strategy if $\alpha^{\dagger} < (1 - \lambda_R)\alpha < \alpha^{\ddagger}$, and a *high-priced add-on* strategy if $\alpha^{\ddagger} < (1 - \lambda_R)\alpha$. Equivalently, as stated in Proposition 2, the firm chooses a *low-priced add-on* strategy if $\alpha < \frac{1}{1 - \lambda_R}\alpha^{\dagger} = \alpha^{\$}$, a price discrimination strategy if $\frac{1}{1 - \lambda_R}\alpha^{\dagger} = \alpha^{\$} < \alpha < \frac{1}{1 - \lambda_R}\alpha^{\ddagger} = \alpha^{\ddagger}$, and a *high-priced add-on* strategy if $\frac{1}{1 - \lambda_R}\alpha^{\ddagger} = \alpha^{\ddagger} < \alpha$. Q.E.D.

A.3 Calculations and proofs for Section 3.2

Proposition 3 (price discrimination and social welfare). The proposition considers the question how a technology that allows firms to price discriminate impacts social welfare. For the calculations, we use that the social welfare loss is equivalent to the number of consumers who substitute away multiplied by the substitution costs e. Note that we can focus the analysis on $\alpha^{\dagger} < \alpha < \alpha^{\ddagger}$. Otherwise, even when a technology to price discriminate is available, a price discrimination strategy is not the optimal price strategy (see Proposition 1). For an overview of results, see Table 1.

First, consider the social welfare loss for $\alpha^{\dagger} < \alpha < \frac{e}{\bar{p}}$. Without price discrimination (in a *low-priced add-on* equilibrium), the social welfare loss is 0. With price discrimination, the

social welfare loss is $(1 - \beta)(1 - \alpha)e$. Thus, the net effect of price discrimination on the social welfare loss is positive $((1 - \beta)(1 - \alpha)e > 0)$. In other words, as stated in the proposition, price discrimination decreases social welfare.

Second, consider the social welfare loss for $\frac{e}{\bar{p}} < \alpha < \alpha^{\ddagger}$. Without price discrimination (in a *high-priced add-on* equilibrium), the social welfare loss is $(1 - \alpha)e$. With price discrimination, the social welfare loss is $(1 - \beta)(1 - \alpha)e$. Thus, the net effect of price discrimination on the social welfare loss is negative $(-\beta(1 - \alpha)e < 0)$. In other words, as stated in the proposition, price discrimination increases social welfare. Q.E.D.

Proposition 4 (price discrimination and prices). The proposition considers the question how a technology that allows firms to price discriminate impacts prices of consumers. Note that we can focus the analysis on $\alpha^{\dagger} < \alpha < \alpha^{\ddagger}$. Otherwise, even when a technology to price discriminate is available, a price discrimination strategy is not the optimal price strategy (see Proposition 1). In the following, we sequentially analyze all relevant cases. For an overview of results, see Tables 2 - 5.

1) Consider prices of sophisticated consumers for $\alpha^{\dagger} < \alpha < \frac{e}{\bar{p}}$. Without price discrimination (under a *low-priced add-on* equilibrium), consumers pay a total for the base good and the add-on of

$$-e + \mu + e = \mu. \tag{19}$$

With price discrimination consumers pay

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \mu + e.$$
⁽²⁰⁾

Thus, the net effect of price discrimination on prices of consumers is

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + e.$$
(21)

According to the proposition, we need to show that the net effect is negative, i.e.,

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + e < 0$$
(22)

$$-e\beta + e\alpha\beta - e\alpha(1-\gamma) - \bar{p}\gamma\alpha + e < 0$$
⁽²³⁾

$$-\alpha(-e\beta + e(1-\gamma) + \bar{p}\gamma) + e(1-\beta) < 0$$
⁽²⁴⁾

$$e(1-\beta) < \alpha(-e\beta + e(1-\gamma) + \bar{p}\gamma)$$
 (25)

$$\frac{e(1-\beta)}{-e\beta + e(1-\gamma) + \bar{p}\gamma} < \alpha$$
(26)

$$\frac{e(1-\beta)}{e(1-\beta) + (\bar{p}-e)\gamma} < \alpha.$$
(27)

Using $\alpha^{\dagger} < \alpha$, where $\alpha^{\dagger} = \min\left(\frac{e}{\bar{p}}, \frac{e(1-\beta)}{e(1-\beta)+(\bar{p}-e)\gamma}\right)$, we know that the inequality holds.¹¹

2) Consider prices of sophisticated consumers for $\frac{e}{\bar{p}} < \alpha < \alpha^{\ddagger}$. Without price discrimination (under a *high-priced add-on* equilibrium), consumers pay a total for the base good and the add-on (more precisely, for the base good and for substituting the add-on) of

$$-\alpha \bar{p} + \mu + e = \mu. \tag{28}$$

With price discrimination consumers pay

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \mu + e.$$
⁽²⁹⁾

Thus, the net effect of price discrimination on prices of consumers is

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \alpha\bar{p}.$$
(30)

¹¹Note that from $\beta \leq (1 - \gamma)$ (see Corollary 1) we know that $(-e\beta + e(1 - \gamma) + \bar{p}\gamma) > 0$, such that dividing with this term does not reverse the direction of the inequality.

According to the proposition, we need to show that the net effect is negative, i.e.,

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \alpha\bar{p} < 0$$
(31)

$$-e\beta + e\alpha\beta - e\alpha(1-\gamma) - \bar{p}\gamma\alpha + \alpha\bar{p} < 0$$
(32)

$$\alpha(e\beta - e(1 - \gamma) - \bar{p}\gamma + \bar{p}) < e\beta$$
(33)

$$\alpha(e\beta - e(1 - \gamma) + \bar{p}(1 - \gamma)) < e\beta$$
(34)

$$\alpha(e\beta + (\bar{p} - e)(1 - \gamma)) < e\beta$$
(35)

$$\alpha < \frac{c\beta}{e\beta + (\bar{p} - e)(1 - \gamma)} \tag{36}$$

Using $\alpha < \alpha^{\ddagger}$, where $\alpha^{\ddagger} = \max\left(\frac{e}{\bar{p}}, \frac{e\beta}{e\beta + (\bar{p} - e)(1 - \gamma)}\right)$, we know that the inequality holds.

3) Consider prices of myopic consumers who are offered the low-priced add-ons for $\alpha^{\dagger} < \alpha < \frac{e}{\bar{p}}$. The analysis is equivalent to the analysis under 1). Thus, we know that the net effect of price discrimination is negative.

4) Consider prices of myopic consumers who are offered the low-priced add-ons for $\frac{e}{\bar{p}} < \alpha < \alpha^{\ddagger}$. Without price discrimination (under a *high-priced add-on* equilibrium), consumers pay a total for the base good and the add-on of

$$-\alpha \bar{p} + \mu + \bar{p}.\tag{37}$$

With price discrimination consumers pay

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \mu + e.$$
(38)

Thus, the net effect of price discrimination on prices of consumers is

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + e - (1-\alpha)\bar{p}.$$
(39)

According to the proposition, we need to show that the net effect is negative, i.e.,

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + e - (1-\alpha)\bar{p} < 0.$$

$$\tag{40}$$

We know from the analysis under 1) that $-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + e < 0$. Thus, we know that the inequality above that differs only by $-(1-\alpha)\bar{p}$ also holds.

5) Consider prices of myopic consumers who are offered the high-priced add-ons for $\alpha^{\dagger} < \alpha < \frac{e}{\bar{p}}$. Without price discrimination (under a *low-priced add-on* equilibrium), consumers pay a total for the base good and the add-on of

$$-e + \mu + e = \mu. \tag{41}$$

With price discrimination consumers pay

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \mu + \bar{p}.$$
(42)

Thus, the net effect of price discrimination on prices of consumers is

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \bar{p}.$$
(43)

According to the proposition, we need to show that the net effect is positive, i.e.,

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \bar{p} > 0$$
(44)

$$-e\beta + e\alpha\beta - e\alpha(1-\gamma) - \bar{p}\gamma\alpha + \bar{p} > 0$$
⁽⁴⁵⁾

$$-\alpha(-e\beta + e(1-\gamma) + \bar{p}\gamma) - e\beta + \bar{p} > 0$$
(46)

$$-e\beta + \bar{p} > \alpha(-e\beta + e(1-\gamma) + \bar{p}\gamma)$$
(47)

$$\frac{\bar{p} - e\beta}{e(1 - \gamma) + \bar{p}\gamma - e\beta} > \alpha$$
(48)

From $\bar{p} > e(1-\gamma) + \bar{p}\gamma$ it follows that $\frac{\bar{p}-e\beta}{e(1-\gamma)+\bar{p}\gamma-e\beta} > 1$. Thus, because $\alpha \in (0,1]$, the inequality above holds.¹²

6) Consider prices of myopic consumers who are offered the high-priced add-ons for $\frac{e}{\bar{p}} < \alpha < \alpha^{\ddagger}$. Without price discrimination (under a *high-priced add-on* equilibrium), consumers pay a

¹²Note that from $\beta \leq (1 - \gamma)$ (see Corollary 1) we know that $(-e\beta + e(1 - \gamma) + \bar{p}\gamma) > 0$, such that dividing with this term does not reverse the direction of the inequality.

total for the base good and the add-on of

$$-\alpha \bar{p} + \mu + \bar{p}.\tag{49}$$

With price discrimination consumers pay

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \mu + \bar{p}.$$
(50)

Thus, the net effect of price discrimination on prices of consumers is

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \alpha\bar{p}.$$
(51)

According to the proposition, we need to show that the net effect is negative, i.e.,

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \alpha\bar{p} < 0.$$
⁽⁵²⁾

We know from the analysis under 2) that inequality above holds.

Next, we consider *expected* prices of myopic consumers. Recall that the *cannibalization* effect of price discrimination implies that firms offer the high-priced add-on to a fraction γ of myopic consumers, and the low-priced add-on (unintentionally) to a fraction $1 - \gamma$ of myopic consumers. From the perspective of a myopic consumer, we interpret these fractions as the probabilities to get offered the high-priced add-on or the low-priced add-on, respectively.

7) Consider *expected* prices of myopic consumers for $\alpha^{\dagger} < \alpha < \frac{e}{\bar{p}}$. Without price discrimination (under a *low-priced add-on* equilibrium), consumers pay a total for the base good and the add-on of

$$-e + \mu + e = \mu. \tag{53}$$

With price discrimination consumers pay *expected* prices of

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \mu + \gamma\bar{p} + (1-\gamma)e.$$
(54)

Thus, the net effect of price discrimination on prices of consumers is

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \gamma\bar{p} + (1-\gamma)e.$$
(55)

According to the proposition, we need to show that the net effect is positive, i.e.,

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \gamma\bar{p} + (1-\gamma)e > 0$$
(56)

$$-e\beta + e\alpha\beta - e\alpha(1-\gamma) - \bar{p}\gamma\alpha + \gamma\bar{p} + (1-\gamma)e > 0$$
(57)

$$-\alpha(-e\beta + e(1-\gamma) + \bar{p}\gamma) > e\beta - \gamma\bar{p} - (1-\gamma)e$$
(58)

$$-\alpha > \frac{e\beta - \gamma \bar{p} - (1 - \gamma)e}{-(e\beta - \gamma \bar{p} - (1 - \gamma)e)}$$
(59)

$$1 > \alpha$$
 (60)

Thus, the inequality holds.¹³

8) Consider *expected* prices of myopic consumers for $\frac{e}{\bar{p}} < \alpha < \alpha^{\ddagger}$. Without price discrimination (under a *high-priced add-on* equilibrium), consumers pay a total for the base good and the add-on of

$$-\alpha \bar{p} + \mu + \bar{p}.\tag{61}$$

With price discrimination consumers pay *expected* prices of

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \mu + \gamma\bar{p} + (1-\gamma)e.$$
(62)

Thus, the net effect of price discrimination on prices of consumers is

$$-e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \gamma\bar{p} + (1-\gamma)e - (1-\alpha)\bar{p}.$$
(63)

¹³Note that from $\beta \leq (1 - \gamma)$ (see Corollary 1) we know that $(-e\beta + e(1 - \gamma) + \bar{p}\gamma) > 0$, such that dividing with this term does not reverse the direction of the inequality.

According to the proposition, we need to show that the net effect is negative, i.e.,

$$\begin{aligned} -e((1-\alpha)\beta + \alpha(1-\gamma)) - \bar{p}\gamma\alpha + \gamma\bar{p} + (1-\gamma)e - (1-\alpha)\bar{p} &< 0 \\ -e\beta + e\alpha\beta - e\alpha(1-\gamma) - \bar{p}\gamma\alpha + \gamma\bar{p} + (1-\gamma)e - \bar{p} + \alpha\bar{p} &< 0 \\ \alpha(e\beta - e(1-\gamma) - \bar{p}\gamma + \bar{p}) &< e\beta - \gamma\bar{p} - (1-\gamma)e + \bar{p} \\ \alpha(e\beta - e(1-\gamma) + \bar{p}(1-\gamma)) &< e\beta + (1-\gamma)\bar{p} - (1-\gamma)e \\ \alpha(e\beta + (\bar{p} - e)(1-\gamma)) &< e\beta + (\bar{p} - e)(1-\gamma) \\ \alpha < 1 \end{aligned}$$

Thus, the inequality holds.

A.4 Calculations and proofs for Section 3.3

Proposition 5 (education and social welfare).This proposition follows directly fromthe calculations of social welfare in Table 6.Q.E.D.

Proposition 6 (education and expected prices). Consider a situation when educating consumers through the regulator cannot change a high-priced add-on pricing equilibrium, i.e., $\alpha^{\sharp} < \alpha$. Without education, the total price for the base good and the add-on that each uninformed consumer pays is

$$p + \hat{p} = -\alpha \bar{p} + \mu + \bar{p}$$
$$= (1 - \alpha)\bar{p} + \mu$$

Recall that a fraction λ_R of uninformed consumers becomes informed through education. This fraction is interpreted as ex-ante probability of an uninformed consumer to become informed. Thus, the total *expected* price that each uninformed consumer pays if the regulator decides

Q.E.D.

to educate consumers is

$$E(p+\hat{p}) = (1-\lambda_R)(-(1-\lambda_R)\alpha\bar{p} + \mu + \bar{p}) + \lambda_R(-(1-\lambda_R)\alpha\bar{p} + \mu + e)$$

$$= (1-\lambda_R)\bar{p} + \lambda_R \ e - (1-\lambda_R)\alpha\bar{p} + \mu$$

$$= (1-\lambda_R)(1-\alpha)\bar{p} + \lambda_R \ e + \mu$$

It follows that educating consumers increases expected prices, if

$$E(p+\hat{p}) > p+\hat{p}$$

$$(1-\lambda_R)(1-\alpha)\bar{p}+\lambda_R \ e+\mu > (1-\alpha)\bar{p}+\mu$$

$$(-\lambda_R)(1-\alpha)\bar{p}+\lambda_R \ e > 0$$

$$(-\lambda_R)(1-\alpha)\bar{p} > -\lambda_R \ e$$

$$(1-\alpha)\bar{p} < e$$

$$\bar{p}-\alpha\bar{p} < e$$

$$\bar{p}-e < \alpha\bar{p}$$

$$1-\frac{e}{\bar{p}} < \alpha$$

Considering the condition $\alpha^{\sharp} < \alpha$ for a *high-priced add-on* pricing strategy, it follows that educating consumers increases expected prices if $\alpha^{\diamondsuit} = \max(1 - \frac{e}{\bar{p}}, \alpha^{\sharp}) < \alpha$. Q.E.D.

A.5 Calculations and proofs for Section 4.2

Proposition 7 (offering of low-priced add-on for all consumers). First, let us consider a *high-priced add-on* equilibrium. Market prices are $p = -\alpha \bar{p} + \mu$ for the base good and $\hat{p} = \bar{p}$ for the add-on. The public firm then offers the base good at the market price and the add-on at *e*. Consider that the regulator offers subsidies to all firms in order to maximize social welfare. Calculations therefore reflect a demand at the public firm of $D(\cdot) = 1$. The expected profit of the public firm per consumer is given by

$$\pi_h = (-\alpha \bar{p} + \mu + e)D(\cdot). \tag{64}$$

Relative to the firms' profits under a high-priced add-on strategy, i.e., μ , the profit is

$$\tilde{\pi}_h = (-\alpha \bar{p} + e) D(\cdot), \tag{65}$$

Recall that a *high-priced add-on* equilibrium exists only for $\alpha > \max\left(\frac{e}{\bar{p}}, \frac{e\beta}{e\beta + (\bar{p}-e)(1-\gamma)}\right)$. Thus, as expected, $\tilde{\pi}_h$ is below zero and represents a relative loss. We define the required subsidy to compensate firms who follow the suggested strategy as

$$S_h = -\tilde{\pi} = (\alpha \bar{p} - e)D(\cdot). \tag{66}$$

The benefit of uninformed consumers is that they each save an amount $\bar{p} - e$. Thus, total redistribution to uninformed consumers is $(\bar{p} - e)\alpha D(\cdot)$. Informed consumers have no benefit because they would pay p + e, i.e., the market price for the base good plus substitution costs e, anyway. Thus, the social welfare gain is the redistribution to uninformed consumers net substitution costs that need to be financed via taxes, i.e.,

$$G_h = (\bar{p} - e)\alpha D(\cdot) - (\alpha \bar{p} - e)D(\cdot)$$
(67)

$$= e(1-\alpha)D(\cdot), \tag{68}$$

which, by construction, equals the prevented substitution costs of informed consumers. Finally, the efficiency of a subsidy is given by

$$E_h = \frac{S_h}{G_h} = \frac{\alpha \bar{p} - e}{e(1 - \alpha)}.$$
(69)

Note that E_h is increasing in α , which is interpreted as the higher the worse. Q.E.D.

A.6 Calculations and proofs for Section 4.3

Proposition 8 (offering of low-priced add-ons for informed consumers). First, let us consider a *high-priced add-on* equilibrium. Market prices are $p = -\alpha \bar{p} + \mu$ for the base good and $\hat{p} = \bar{p}$ for the add-on. The public firm then offers the base good above the equilibrium market price, i.e., $p + \Delta$, and the add-on for $\hat{e} - \Delta$, where $\hat{e} < e$. Consider that the regulator offers subsidies to all firms in order to maximize social welfare. Because the public firm will only have informed customers, the total demand at the public firms is $D(\cdot) = 1 - \alpha$, and the total demand at the private firms is $1 - D(\cdot) = \alpha$.

Because private firms only have uninformed consumers, the market price of the base good decreases from $p = -\alpha \bar{p} + \mu$ to $p = -\bar{p} + \mu$. The expected profit of the public firm is given by

$$\pi_h = (-\bar{p} + \mu + \Delta + \hat{e} - \Delta)D(\cdot). \tag{70}$$

Relative to the firms' profits under a high-priced add-on strategy, i.e., μ , the profit is

$$\tilde{\pi}_h = (-\bar{p} + \hat{e})D(\cdot),\tag{71}$$

Recall that a *high-priced add-on* equilibrium exists only for $\alpha > \max\left(\frac{e}{\bar{p}}, \frac{e\beta}{e\beta + (\bar{p}-e)(1-\gamma)}\right)$. Thus, as expected, $\tilde{\pi}_h$ is below zero and represents a relative loss. We define the required subsidy to compensate firms who follow the suggested strategy as

$$S_h = -\tilde{\pi} = (\bar{p} - \hat{e})D(\cdot)$$

$$= (\bar{p} - \hat{e})(1 - \alpha).$$
(72)

Note that the expected profit of private firms is unchanged. They yield

$$\pi = (-\bar{p} + \mu + \bar{p})D(\cdot)$$

$$= \mu D(\cdot).$$
(73)

The benefit of uninformed consumers is that they each save an amount $(1 - \alpha)\bar{p}$ because of the lower price of the base good. The benefit of informed consumers is twofold: first, $(1 - \alpha)\bar{p}(1 - \alpha)$ because they get the base good at a lower price, and second, $(e - \hat{e})(1 - \alpha)$ because they get the add-on below their substitution costs. The resulting social welfare gain is the redistribution to uninformed and informed consumers net substitution costs that need to be financed via taxes, i.e.,

$$G_{h} = (1-\alpha)\bar{p}(1-D(\cdot)) + ((1-\alpha)\bar{p} + (e-\hat{e}))D(\cdot) - (\bar{p}-\hat{e})D(\cdot)$$
(74)
$$= (1-\alpha)\bar{p}\alpha + ((1-\alpha)\bar{p} + (e-\hat{e}))(1-\alpha) - (\bar{p}-\hat{e})(1-\alpha)$$

$$= e(1-\alpha),$$

which, by construction, equals the prevented substitution costs of informed consumers. Finally, the efficiency of a subsidy is given by

$$E_h = \frac{S_h}{G_h} = \frac{\bar{p} - e}{e}.$$
(75)

Note that E_h is constant in α .

A.7 Calculations and proofs for Section 4.5

Proposition 9 (subsidizing firms). Under a high-priced add-on equilibrium, subsidizing firms that offer low-priced add-ons to all consumers requires subsidies of $\alpha \bar{p} - e$ to reach a net social welfare gain of $e(1 - \alpha)$, as shown in Proposition 7. Subsidizing firms that offer low-priced add-ons only to informed consumers requires subsidies of $(\bar{p} - \hat{e})(1 - \alpha)$ for the same net social welfare gain, as shown in Proposition 8. Thus, the former strategy requires lower subsidies, if

$$\alpha \bar{p} - e < (\bar{p} - \hat{e})(1 - \alpha) \tag{76}$$

$$\alpha \bar{p} - e < \bar{p} - \bar{p}\alpha - \hat{e} + \hat{e}\alpha \tag{77}$$

$$\alpha \bar{p} - \hat{e}\alpha + \bar{p}\alpha < \bar{p} - \hat{e} + e \tag{78}$$

$$\alpha < \frac{\bar{p} - \hat{e} + e}{\bar{p} - \hat{e} + \bar{p}}.$$
(79)

Q.E.D.

Q.E.D.

B Appendix: Figures

Figure 1: Price discrimination

This figure shows effects of price discrimination on consumers without regulatory intervention. The fractions of informed and uninformed consumers (pre cannibalization) are $1 - \alpha$ (sophisticated consumers) and α (myopic consumers), respectively. The *attrition effect* reflects that the firm will unintentionally *shroud* (not advertise) the low-priced add-on towards a fraction $1 - \beta$ of informed consumers who consequently substitute away. The *cannibalization effect* reflects that the firm will unintentionally *unshroud* (advertise) the low-priced add-on towards a fraction $1 - \gamma$ of myopic consumers, who consequently buy the low-priced add-on instead of the high-priced add-on. Note that a fraction λ of these consumers then becomes informed and a fraction $1 - \lambda$ stays uninformed. This is, however, not relevant because these consumers all buy the low-priced add-on they get offered anyway.



Figure 2: Threshold level for α without regulatory intervention

The firm chooses a *low-priced add-on* strategy if $\alpha < \alpha^{\dagger}$, a *price discrimination* strategy if $\alpha^{\dagger} < \alpha < \alpha^{\ddagger}$, and a *high-priced add-on* strategy if $\alpha^{\ddagger} < \alpha$. The figure illustrates a case where price discrimination is efficient $(\beta > 1 - \gamma)$.



Figure 3: Threshold level for α with regulatory intervention

Regulatory intervention change the threshold level by $\frac{1}{1-\lambda}$ compared with the base case without regulatory intervention. The threshold level that determine the optimal pricing strategy of a firm are defined as $\alpha^{\S} = \frac{1}{1-\lambda}\alpha^{\dagger}$ and $\alpha^{\sharp} = \frac{1}{1-\lambda}\alpha^{\ddagger}$. The firm chooses a *low-priced add-on* strategy if $\alpha < \alpha^{\S}$, a *price discrimination* strategy if $\alpha^{\S} < \alpha < \alpha^{\sharp}$, and a *high-priced add-on* strategy if $\alpha^{\sharp} < \alpha$. The figure illustrates a case where price discrimination is efficient ($\beta > 1 - \gamma$).





Figure 4: price discrimination and social welfare

The figure illustrates *social welfare loss* for varying α with price discrimination and with uniform pricing.

Suppose price discrimination is not fully efficient ($\beta = 0.7$ and $\gamma = 0.7$), $\bar{p} = 1$ and e = 0.3.



Suppose price discrimination is not fully efficient ($\beta = 0.7$ and $\gamma = 0.7$), $\bar{p} = 1$ and e = 0.3.



Figure 6: education and social welfare loss (with price discrimination)

The figure illustrates the *social welfare loss* for varying α , with and without education. Suppose educating consumers makes 30% of myopic consumers informed.



Figure 7: education and social welfare loss (price discrimination not efficient)

Let e = 0.25, and $\bar{p} = 1$. The social welfare loss without regulatory intervention is shown as dotted line, and the social welfare loss with regulatory intervention is shown as solid line.



(a) varying α and $\lambda = 0.25$: Without regulatory intervention, a social welfare loss comes from a shrouded price equilibrium for $\alpha > 0.25$. With regulatory intervention, a social welfare loss comes from a shrouded price equilibrium for $\alpha > 0.33$.

(b) varying λ and $\alpha = 0.75$:

Figure 8: education and social welfare loss (price discrimination w/o attrition)

Let $\beta = 1$, $\gamma = 0.2$, e = 0.25, and $\bar{p} = 1$. The social welfare loss without regulatory intervention is shown as dotted line. The social welfare loss is shown as solid line.



(a) varying α and $\lambda = 0.25$: Without regulatory intervention, a social welfare loss comes from a shrouded price equilibrium with uniform pricing for $\alpha > 0.625$. With regulatory intervention, a social welfare loss comes from a shrouded price equilibrium with uniform pricing for $\alpha > 0.833$.

(b) varying λ and $\alpha = 0.75$:

Figure 9: education and prices

Suppose educating consumers makes 30% of myopic consumers informed.



Figure 10: pitfalls of the regulator

Suppose the regulator observes *high-priced add-ons*, and estimates a fraction of myopic consumers $\alpha = 0.6$. Furthermore, the regulator estimates that educating consumers makes a fraction $\lambda = 0.3$ informed. however, the regulator can only observe the status quo and does not know whether a technology for price discrimination is available to firms. Thus, a situation as shown in Figure a) may be relevant where educating consumers decreases social welfare, or a situation as shown in Figure b) where educating consumers increases social welfare.



Figure 11: Cash flows under alternative price equilibria

The figure illustrates the cash flows under alternative price equilibria that depend on the fraction of myopic consumers α . Note that under all alternatives, the total profit of private firms is μ . Consumers always consume the base good and the add-on (or alternatively the substitution good). If some consumers substitute away at costs e, this leads to higher prices of the base good which harms all consumers. For example, substitution costs (high-priced add-on) = total prices of all consumers (high-priced add-on) - total prices of all consumers (low-priced add-on) = social welfare loss.

Note that, for clarity, each term in the figure begins with the economic effect on an affected consumer, and ends with the fraction of affected consumers in square brackets.



Figure 12: Offering of low-priced add-ons to all consumers under a *high-priced add-on* equilibrium

The figure shows cash flows relative to a high-priced add-on equilibrium without a public firm. A "public firm" represents a firm that accepts to follow a certain (loss-making) pricing strategy suggested by the regulator, and in turn receives a subsidy from the regulator. Suppose that the suggested strategy is to offer the base good for the equilibrium market price, i.e., $p = -\alpha \bar{p} + \mu$, and the add-on for e. The total volume of potential subsidies is such that all firms can participate. The suggested pricing strategy has two consequences for firms: First, they make an additional profit of $e[1 - \alpha]$ with all informed consumers who otherwise substitute away in a high-priced add-on equilibrium. Second, the firms' profits with uninformed consumers decline by $(\bar{p} - e)[\alpha]$. In sum, this pricing strategy is loss-making for firms. Thus, a subsidy needs to balance out the net effect on firms of $\alpha \bar{p} - e$, otherwise firms would not accept the suggested pricing strategy. This subsidy, of course, needs to be financed by consumer through taxes. It follows that the net effect on all consumers is $(\bar{p} - e)\alpha - \alpha \bar{p} - e = e(1 - \alpha)$ which, by construction, equates to the prevented substitution costs of informed consumers.



Figure 13: Offering of low-priced add-ons only to informed consumers under a *high-priced* add-on equilibrium

The figure shows cash flows relative to a *high-priced add-on* equilibrium without a public firm. A "public firm" represents a firm that accepts to follow a certain (loss-making) pricing strategy suggested by the regulator, and in turn receives a subsidy from the regulator. Suppose that the suggested strategy is to offer the base good above the equilibrium market price, i.e., $p + \Delta$, and the add-on for $\hat{e} - \Delta$, where $\hat{e} < e$. The total volume of potential subsidies is such that firms who represent $1 - \alpha$ of the market can participate.

Then, all informed consumers who otherwise would buy at private firms migrate to the public firm, and all uninformed consumers who otherwise buy at the public firm migrate to the private firms. Equilibrium market prices for the base good decrease from $-\alpha \bar{p} + \mu$ to $-\bar{p} + \mu$. Therefore, existing uninformed consumers at private firms, i.e., a fraction $\alpha \alpha$ of all consumers, and existing informed consumers at public firms, i.e., a fraction $(1 - \alpha)(1 - \alpha)$ of all consumers, gain $(1 - \alpha)\bar{p}$.

Furthermore, existing informed consumers at public firms buy the add-on at the public firm for \hat{e} and do not substitute away. Uninformed consumer who migrate to private firms decrease profits at public firms by $(-\alpha \bar{p} + \mu + \bar{p})[\alpha(1-\alpha)]$ and increase profits at private firms by

 $(-\bar{p} + \mu + \bar{p})[\alpha(1 - \alpha)]$. Informed consumer who migrate to public firms decrease profits at private firms by $(-\alpha \bar{p} + \mu)[(1 - \alpha)\alpha]$ and increase profits at public firms by $(-\bar{p} + \mu + \hat{e})[(1 - \alpha)\alpha]$. The new informed consumers at the public firm do not substitute away. In total, substitution costs for existing and new informed consumers decrease by $e[1 - \alpha]$.

In sum, there are no effects on profits of private firms, but negative effects on profits of public firms. Thus, a subsidy of $(\bar{p} - \hat{e})[1 - \alpha]$ needs to balance out the net effects on public firms, otherwise firms would not accept the suggested pricing strategy. This subsidy, of course, needs to be financed by consumer through taxes. The net effect on uninformed consumers (excluding taxes) is a gain of $(1 - \alpha)\bar{p}[\alpha]$. The net effect on informed consumers (excluding taxes) is a gain of $((1 - \alpha)\bar{p} + (e - \hat{e}))[1 - \alpha]$. It follows that the net effect on all consumers including taxes is $(1 - \alpha)\bar{p}[\alpha] + ((1 - \alpha)\bar{p} + (e - \hat{e}))[1 - \alpha] - (\bar{p} - \hat{e})[1 - \alpha] = e[1 - \alpha]$ which, by construction, equates to

the prevented substitution costs of informed consumers. Note that, for clarity, each term in the figure begins with the economic effect on an affected consumer, and ends with the fraction of affected consumers in square brackets.



Figure 14: Required subsidies under alternative regulatory strategies

The Figure illustrates a situation without education where price discrimination is not attractive $(\bar{p} = 1, e = 0.3, \hat{e} = 0.3, \beta = 0)$. Both alternative regulatory strategies, offering low-priced add-ons to *all* consumers or offering low-priced add-ons only to *informed* consumers, yield a net social welfare gain of $e[1 - \alpha]$. Thus, the regulator will choose the strategy that requires the least subsidies (taxes). For this numerical example, the optimal strategy is offering low-priced add-ons to *all* consumers for $0.4 < \alpha < 0.65$ and offering low-priced add-ons only to *informed* consumers for $0.65 < \alpha$.



C Appendix: Tables

Cells without any shading represent a *low-priced add-on* equilibrium. Cells with a light grey shading represent a *price discrimination* equilibrium. Cells with a dark grey shading represent a *high-priced add-on* equilibrium. The thresholds $(\alpha^{\dagger}, \alpha^{\ddagger}, \alpha^{\$} \text{ and } \alpha^{\ddagger})$ are defined in Proposition 1 and Proposition 2.

Table 1: Social welfare loss and price discrimination (w/o education) Social welfare loss comes from informed consumers who substitute away at costs e if the firm does not offer them the add-on for a price of at most e. The first row shows the social welfare loss without price discrimination, the second row shows the social welfare loss with price discrimination, the third row shows the resulting net effect of price discrimination (second row minus first row), and the fourth row shows the sign of the net effect.

$\alpha < \alpha^\dagger$	$ \alpha^{\dagger} < \alpha < \frac{e}{\bar{p}} $	$\tfrac{e}{\bar{p}} < \alpha < \alpha^{\ddagger}$	$\alpha^{\ddagger} < \alpha$
0	0	$(1-\alpha)e$	$(1-\alpha)e$
0	$(1-\beta)(1-\alpha)e$	$(1-\beta)(1-\alpha)e$	$(1-\alpha)e$
0	$ (1-\beta)(1-\alpha)e $	$-\beta(1-\alpha)e$	0
	(+)	(-)	

Table 2: Prices for sophisticated consumers and price discrimination The table shows prices for the base good plus the add-on under alternative price equilibria for sophisticated consumers. The first row shows prices without price discrimination, the second row shows prices with price discrimination, the third row shows the resulting net effect of price discrimination (second row minus first row), and the fourth row shows the sign of the net effect.

$\alpha < \alpha^{\dagger}$	$\alpha^{\dagger} < \alpha < \frac{e}{\bar{p}}$	$\frac{e}{\bar{p}} < \alpha < \alpha^{\ddagger}$	$\alpha^{\ddagger} < \alpha$
$-e + \mu + e = \mu$	$-e + \mu + e = \mu$	$-\alpha \bar{p} + \mu + e$	$-\alpha \bar{p} + \mu + e$
$-e + \mu + e = \mu$	$-e((1-\alpha)\beta + \alpha(1-\gamma))$	$-e((1-\alpha)\beta + \alpha(1-\gamma))$	$-\alpha \bar{p} + \mu + e$
	$-\bar{p}\gamma\alpha + \mu + e$	$-\bar{p}\gamma\alpha + \mu + e$	
0	$-e((1-\alpha)\beta + \alpha(1-\gamma))$	$-e((1-\alpha)\beta + \alpha(1-\gamma))$	0
	$-\bar{p}\gamma\alpha + e$	$-\bar{p}\gamma\alpha + \alpha\bar{p}$	
	(-)	(-)	

Table 3: Prices for myopic consumers and price discrimination with low-priced add-ons The table shows total prices for the base good plus the add-on under alternative price equilibria for myopic consumers. Here, the myopic consumers (unintentionally) get offered the low-priced add-ons in the *price discrimination* equilibrium, reflecting what we call *cannibalization*. The first row shows prices without price discrimination, the second row shows prices with price discrimination, the third row shows the resulting net effect of price discrimination (second row minus first row), and the fourth row shows the sign of the net effect.

$\alpha < \alpha^{\dagger}$	$ \qquad \alpha^{\dagger} < \alpha < \frac{e}{\bar{p}}$	$\frac{e}{\bar{p}} < \alpha < \alpha^{\ddagger}$	$ \alpha^{\ddagger} < \alpha$
$-e + \mu + e = \mu$	$ \qquad -e + \mu + e = \mu$	$-\alpha \bar{p} + \mu + \bar{p}$	$-\alpha \bar{p} + \mu + \bar{p}$
$-e + \mu + e = \mu$	$-e((1-\alpha)\beta + \alpha(1-\gamma))$	$-e((1-\alpha)\beta + \alpha(1-\gamma))$	$-\alpha \bar{p} + \mu + \bar{p}$
	$-p\gamma\alpha+\mu+e$	$-p\gamma\alpha+\mu+e$	
0	$-e((1-\alpha)\beta + \alpha(1-\gamma))$	$-e((1-\alpha)\beta + \alpha(1-\gamma))$	0
	$-p\gamma\alpha + e$	$-p\gamma\alpha + e - (1 - \alpha)p$	
	(-)	(-)	

Table 4: Prices for myopic consumers and price discrimination with high-priced add-ons The table shows prices for the base good plus the add-on under alternative price equilibria for myopic consumers. Here, the myopic consumers get offered the high-priced add-ons in the *price discrimination* equilibrium. The first row shows prices without price discrimination, the second row shows prices with price discrimination, the third row shows the resulting net effect of price discrimination (second row minus first row), and the fourth row shows the sign of the net effect.

$\alpha < \alpha^{\dagger}$	$ \qquad \alpha^{\dagger} < \alpha < \frac{e}{\bar{p}}$	$\left \frac{e}{\bar{p}} < \alpha < \alpha^{\ddagger} \right $	$ \qquad \alpha^{\ddagger} < \alpha$
$-e + \mu + e = \mu$	$ \qquad -e + \mu + e = \mu$	$-\alpha \bar{p} + \mu + \bar{p}$	$-\alpha \bar{p} + \mu + \bar{p}$
$-e + \mu + e = \mu$	$-e((1-\alpha)\beta + \alpha(1-\gamma))$	$-e((1-\alpha)\beta + \alpha(1-\gamma))$	$-\alpha \bar{p} + \mu + \bar{p}$
	$-p\gamma\alpha+\mu+p$	$-p\gamma\alpha+\mu+p$	
0	$ -e((1-\alpha)\beta+\alpha(1-\gamma)) $	$ -e((1-\alpha)\beta+\alpha(1-\gamma)) $	0
	$-\bar{p}\gamma\alpha+\bar{p}$	$-\bar{p}\gamma\alpha + \alpha\bar{p}$	
	(+)	(-)	

Table 5: Expected prices for myopic consumers and price discrimination The table shows *expected* prices for the base good plus the add-on under alternative price equilibria for myopic consumers. Recall that the *cannibalization* effect of price discrimination implies that firms offer the high-priced add-on to a fraction γ of myopic consumers, and the low-priced add-on (unintentionally) to a fraction $1 - \gamma$ of myopic consumers. From the perspective of a myopic consumer, we interpret these fraction as the probabilities to get offered the high-priced add-on or the low-priced add-on, respectively. The first row shows prices without price discrimination, the second row shows expected prices with price discrimination, the third row shows the resulting net effect of price discrimination (second row minus first row), and the fourth row shows the sign of the net effect.

$\alpha < \alpha^{\dagger}$	$ \qquad \alpha^{\dagger} < \alpha < \frac{e}{\bar{p}}$	$rac{e}{ar{p}} < lpha < lpha^{\ddagger}$	$\alpha^{\ddagger} < \alpha$
$-e + \mu + e = \mu$	$ \qquad -e + \mu + e = \mu$	$-\alpha \bar{p} + \mu + \bar{p}$	$-\alpha \bar{p} + \mu + \bar{p}$
$-e+\mu+e=\mu$	$-e((1-\alpha)\beta + \alpha(1-\gamma))$	$-e((1-\alpha)\beta + \alpha(1-\gamma))$	$-\alpha \bar{p} + \mu + \bar{p}$
	$ -p\gamma\alpha + \mu + \gamma p + (1 - \gamma)e$	$-p\gamma\alpha + \mu + \gamma p + (1 - \gamma)e$	
0	$-e((1-\alpha)\beta + \alpha(1-\gamma))$	$-e((1-\alpha)\beta + \alpha(1-\gamma))$	0
	$\left -\bar{p}\gamma\alpha + \gamma\bar{p} + (1-\gamma)e \right $	$-\bar{p}\gamma\alpha + \gamma\bar{p} + (1-\gamma)e - (1-\alpha)\bar{p}$	
	(+)	(-)	

$\alpha < \alpha^\dagger$	$ \alpha^{\dagger} < \alpha < \alpha$	$< \alpha^{\S}$	$\alpha^{\S} < \alpha < \alpha^{\ddagger}$		$\alpha^{\ddagger} < \alpha < \alpha^{\sharp}$	$\alpha^{\sharp} < \alpha$
0	$ $ $(1-\beta)(1$	$-\alpha)e$	$(1-\beta)(1-\alpha)e$		$(1-\alpha)e$	$(1-\alpha)e$
0	0	($(1-\beta)(1-(1-\lambda_R)\alpha)$)e	$(1-\beta)(1-(1-\lambda_R)\alpha)e$	$(1-(1-\lambda_R)\alpha)e$
0	$ -(1-\beta)(1-\beta)(1-\beta) $	$-\alpha)e$	$(1-\beta)\lambda_R \alpha e$	-	$-(\beta(1-\alpha)-(1-\beta)\lambda_R\alpha)e$	$\lambda_R \alpha e$
	(-)		(+)		(-)	(+)

Table 6: Social welfare loss and education Social welfare loss comes from informed consumers who substitute away at costs e if the firm does not offer

them the add-on for a price of at most e. The first row shows the social welfare loss without education, the second row shows the social welfare loss with education, the third row shows the resulting net effect of

education (second row minus first row), and the fourth row shows the sign of the net effect.

Table 7: Fraction of exploited consumers

The first row shows the social welfare loss without education, the second row shows the social welfare loss with education, the third row shows the resulting net effect of education, and the fourth row shows the sign of the net effect.

$\alpha < \alpha^\dagger$	$\left \begin{array}{c} \alpha^{\dagger} < \alpha < \alpha^{\S} \end{array} \right.$	$ \alpha^{\S} < \alpha < \alpha^{\ddagger}$	$\left \alpha^{\ddagger} < \alpha < \alpha^{\sharp} \right.$	$ \alpha^{\sharp} < \alpha$
0	$(1-\gamma)\alpha$	$(1-\gamma)\alpha$	α	α
0	0	$ (1-\gamma)(1-\lambda)\alpha$	$ (1-\gamma)(1-\lambda)\alpha $	$(1-\lambda)\alpha$
0	$ -(1-\gamma)\alpha$	$ -(1-\gamma)\lambda\alpha$	$ -(\lambda + \gamma - \lambda \gamma)$	$ -\lambda \alpha$
	(-)	(-)	(-)	(-)